### **Canterbury District Health Board's**

### POSITION STATEMENT ON ALCOHOL

This position statement is consistent with the position statements of Nelson Marlborough, West Coast, Canterbury, South Canterbury, and Southern District Health Boards<sup>1</sup> and should be read in conjunction with the evidence-based background paper on alcohol.<sup>2</sup> Both documents have been developed collaboratively by the South Island Public Health Units and represent the South Island DHBs working together to address alcohol-related harm.

The Canterbury District Health Board acknowledges the wide range of alcohol-related harm that is experienced by people within the Canterbury district and that the burden of this harm is carried disproportionately by some population groups. It recognises that alcohol use is a major risk factor for numerous health conditions, injuries and social problems. Additionally, alcohol-related harm costs the health sector significant money, time and resources.

### **CANTERBURY DHB POSITION:**

The Canterbury District Health Board will reduce the alcohol-related harm experienced by people within the Canterbury district by developing an Alcohol Harm Reduction Strategy. This strategy will set out the actions Canterbury District Health Board will undertake to reduce alcohol-related harm, including a communication plan.

The Canterbury District Health Board will identify and record alcohol-related presentations within the Canterbury district in a consistent manner.

The Canterbury District Health Board will support and assist Territorial Authorities to develop local alcohol plans that seek to reduce alcohol-related harm by providing information on alcohol-related presentations to emergency departments, and other information pertaining to the burden of alcohol. It will provide further evidence-based advice to assist with these plans.

### **EVIDENCE BASED SOLUTIONS:**

The Canterbury District Health Board will advocate for the following evidence-based solutions to reduce the alcohol-related harm experienced by New Zealanders<sup>3</sup>:

<sup>&</sup>lt;sup>1</sup> Individual DHBs: delete own DHB from this list as appropriate.

<sup>&</sup>lt;sup>2</sup> A summary of evidence from the paper is attached as an appendix. Full references are in the background paper.

<sup>&</sup>lt;sup>3</sup> These recommendations align with the CDHB's Submission to The Law Commission's Issues Paper on the Reform of New Zealand's Liquor Laws (2009), and with those contained in a recent

Commentary from the Injury Prevention Research Unit: Kypri, K., Maclennan, B., Langley, J.D., and Connor, J.L. 2011. 'The *Alcohol Reform Bill*: More tinkering than reform in response to the New Zealand public's demand for better laws'. *Drug and Alcohol Review* 30, 428-433.

### **Raise alcohol prices**

- Increase levels of excise tax on alcohol by at least 50%
- Adjust excise tax so that alcohol products taxed directly on level of ethanol
- Use revenue from increase in excise tax to reduce harm amongst high-risk consumers
- Set minimum retail price for alcohol (per alcohol unit)

### Raise the alcohol purchase age

- Restore alcohol purchase age to 20 years for both on-licences and off-licences
- Ensure enforcement of minimum purchase age
- Additionally, make it an offence for an adult other than a parent/guardian to supply alcohol to a child; and require parents/guardians who supply alcohol to their child to supervise the consumption of that alcohol

### **Reduce alcohol accessibility**

- Restrict on-licences from selling alcohol after 2am
- Restrict off-licences to selling alcohol between 8am and 10pm
- Restrict convenience stores / dairies from selling alcohol
- Tighten law on granting of liquor licences provide further grounds to refuse licences (e.g. detrimental social impact to community)
- Tighten restrictions on numbers of outlets in a given area

### Reduce marketing and advertising of alcohol

- Ban alcohol sponsorship of sporting and cultural events
- Ban advertising of alcohol from television and cinema
- Advertising of alcohol to convey only basic information about the product
- Put health warning labels on alcohol products
- Ensure alcoholic beverages are labelled with ingredient and nutritional information
- Prohibit marketing of alcohol to youth

### Reduce legal blood-alcohol limits for drivers

• Lower the legal blood alcohol (BAC) limit from 80mg/100ml blood to 50mg/100ml blood

### **APPENDIX:** Summary of Evidence

### **Alcohol Related Harm:**

Alcohol use is a major risk factor for numerous health conditions, injuries and social problems, causing approximately 4% of deaths worldwide and (in 2000) 3.9% of all deaths in New Zealand. Much acute harm results from intoxication and includes: road traffic injuries and fatalities, burns, falls, drowning, poisoning, foetal alcohol spectrum disorder, assault, self-inflicted injury, suicide and homicide.

### **Biological effects of alcohol**

Alcohol affects the brain. It alters the mood and impairs memory and psychomotor function. People who consume alcohol are less inhibited and therefore more likely to take risks and behave aggressively, leading to motor vehicle accidents and other injuries. Alcohol use is linked to a wide range of major diseases, including: heart disease, cancer, psychiatric and neurological conditions, gastrointestinal disease, and birth defects including foetal alcohol syndrome. It also contributes to diabetes, sleep disorders, and infectious diseases such as pneumonia and tuberculosis.

Unborn children and adolescents are particularly vulnerable to the effects of alcohol. Unborn children exposed to alcohol are at high risk of problems with memory, language, attention, learning, Visio-spatial ability, fine and gross motor skills, and social and adaptive functioning. Adolescent brains are still developing and therefore vulnerable to alcohol toxicity, addictive problems and psychiatric disorders.

### Alcohol-related harm

Alcohol contributes to crime in New Zealand. Nearly half of all homicides in New Zealand between 1999 and 2008 involved alcohol. A third of all offenders in the year 2007/08 had consumed alcohol. Drink driving causes substantial harm - 27% of drivers in all fatal crashes between 2007 and 2009 were reported as having consumed alcohol.

Social harm results from alcohol: reportedly 12.2% of adults experienced harmful effects on friendships, social life, home life, work/study/employment opportunities, financial position, and legal problems or difficulty learning from their own drinking in the past year.

The economic cost of alcohol-related harm in New Zealand is significant. Harmful alcohol use in 2005/06 alone cost New Zealand an estimated \$4,794 million of diverted resources and lost welfare.

### Alcohol-related harm and population groups

Alcohol-related harm is experienced variably throughout the population. Men have a higher rate of alcohol-related mortality than women and Māori have a higher rate than non-Māori. Evidence clearly demonstrates that Māori suffer disproportionately from a wide range of alcohol-related harms compared to non-Māori. New Zealanders with lower socioeconomic status also bear a disproportionate burden of alcohol-related harm. Children are particularly vulnerable to alcohol-related harm caused by the drinking of other people and can suffer from increased susceptibility to child abuse, neglect and witnessing family violence if caregivers have an alcohol problem.

### Cost of alcohol-related harm to the health sector

Alcohol-related harm in New Zealand costs the health sector significant money, time and resources. Intoxicated patients also impact negatively on staff and other patients. An estimated 35% of injury-based emergency department presentations are alcoholrelated. From 1 November 2010 to 29 October 2011 892 patients were seen in Dunedin Hospital Emergency Department for alcohol-related presentations. The average length of stay for these patients was 4.5 hours, with an average cost to Southern District Health Board of \$1,000 per person.

### NZ Drinking Pattern:

### Alcohol is widely available in NZ

Alcohol is easily accessible from a wide variety of outlets and to anyone over the age of 18. It can be purchased 24 hours a day, 7 days a week and on most days of the year. Alcohol can be consumed either on the premises (on-licences) in bars, restaurants, cafes, hotels, pubs and individual clubs or at special functions; or off the premises (off-licences) when purchased from liquor stores, supermarkets, grocery stores or dairies. Alcohol is more widely available now than in the past: in 2010 the number of places which held liquor licences was 14,424; this has increased from 6,295 in 1990. It is inexpensive: reportedly, in 2010, 3 litres of cask wine could be purchased (on special) for as little as \$16.99.

### Drinking patterns in NZ

According to recent surveys, most New Zealanders (85%) drink at least some alcohol. At least two-thirds of those surveyed in 2007/08 drank once a week. Of people surveyed, nearly two-thirds of all people drank to excess at least once a year and one in ten did so at least once a week. Harmful drinking is more common amongst Māori, Pacific and young people. New Zealanders tolerate excess drinking – less than half surveyed agreed that "It is never O.K. to get drunk" and over one quarter agreed that it is "O.K. to get drunk as long as it's not everyday". A third of those surveyed started drinking at around the age of 14.

### How the current law impacts upon these drinking patterns

The Sale of Liquor Act (1989) has liberalised the sale of alcohol, allowing it to be sold widely, including from supermarkets and over a 24 hour period. Since 1999 (with an amendment to the Act), the purchase age has dropped to 18 (from 20 years), beer has become available in supermarkets and alcohol can be purchased on Sundays. District Licensing Authorities (DLAs) in each local area grant and renew licenses and stipulate opening times. Licensing Inspectors check that premises within their area comply with regulations (e.g. not selling to those who are already intoxicated). The Resource Management Act (1991) legislates how local communities manage the use of land, which requires that a District Plan be put into place and complied with. The Local Government Amendment Act 2001 allows local authorities to impose liquor bans, banning alcohol in public places at certain times. The Land Transport Amendment Act (2011) has lowered the blood alcohol concentration (BAC) limit for drivers under 20 years to zero. The limit for drivers over 20 years is 80mg per 100ml blood.

### **Evidence Based Strategies to Reduce Harm:**

#### **Raise prices**

Evidence shows that when alcohol prices go up, consumption goes down. One of the best ways to influence the consumption of alcohol is through pricing. Alcohol prices are subject to excise tax, which in New Zealand is set at a particular rate depending on which band of alcohol strength the product falls into (e.g. alcoholic beverages between 9-14% alcohol are taxed at 10%). Currently excise tax rates are lower than that of other countries; they are also not adjusted for inflation. In New Zealand there is often a price differential between on and off-licences, which encourages 'preloading' (loading up on cheap alcohol before frequenting on-licences).

#### Raise the purchase age

Research shows that the legal purchase age affects how much youth drink. A lower purchase age has been associated with increased harm (including traffic crashes). In order for a higher purchase age to be effective, it needs to be combined with adequate enforcement. A higher purchase age acknowledges that the effect of alcohol and its harms is much greater on the adolescent brain as it is still developing.

#### **Reduce alcohol accessibility**

It is scientifically and economically effective to restrict the physical availability of alcohol in order to reduce harm. Limiting the physical availability of alcohol can be achieved through limiting the hours and days of sale, and controlling outlet density. Currently alcohol is too easily purchased and facilitates pre-loading. There are often too many alcohol outlets within an area – high densities of alcohol outlets have been shown to be associated with increased harm, including traffic crashes.

### Reduce marketing and advertising

Advertising of alcohol has increased in many countries over recent decades, including New Zealand. Prior to the 1980s alcohol advertising in New Zealand was mostly non-existent, due to legislation controlling the advertising of alcohol – now alcohol advertising is left to the self-regulation of the industry. Since 1992, advertising of alcohol has been allowed on both television and radio – albeit at restricted times (9pm-6am) for television. Since 1987 alcohol companies have been allowed to sponsor sports and advertise corporately. Alcohol advertisements often sell the image that drinking is attractive, glamorous and fun; and these messages are particularly appealing to young people. Alcohol advertising not only leads to greater consumption of alcohol, but also colours people's perceptions of the drinking habits of others.

### **Reduce legal blood alcohol limits for drivers**

With increasing levels of alcohol in the blood, driving performance declines. Currently (as of 2011), there is zero tolerance for drivers under 20 years with any alcohol at all in their blood. Drivers over 20 are legally entitled to drive after drinking with no more than 80mg per 100ml of alcohol in the blood. In 2009 in New Zealand, 138 deaths resulted from traffic accidents where alcohol (and/or drug use) was a contributing factor. Research has shown that the risk of traffic crashes goes up proportionate to the level of alcohol in the blood: the risk doubles for those with 0.05% BAC compared to those with none; there is ten times the risk for those with 0.08% BAC; and one hundred times the risk for those with 0.15% BAC or higher.

# **BACKGROUND PAPER**

# of Supporting Evidence to the South Island District Health Boards'

# **POSITION STATEMENT on ALCOHOL**

**Prepared by:** 

Willow MacDonald, Public Health South, Southern DHB Sarah Colhoun and Susan Bidwell, Community and Public Health, Canterbury DHB

### **Executive summary**

- This background paper provides supporting evidence to the accompanying Position Statement. It outlines the current legislation and regulations relating to alcohol in New Zealand, followed by the effects of alcohol on personal health, the social and economic harms related to alcohol, and approaches to reduce alcohol-related harm.
- The quantity of alcohol available for consumption, the number and variety of locations where alcohol is sold in both on- and off-licensed premises, and the relatively cheap price mean that alcohol is widely available in New Zealand. Additionally, there are almost no restrictions on the hours of the day or days of the year on which alcohol may be sold.
- The main piece of legislation controlling alcohol in New Zealand is the Sale of Liquor Act (1989) and its subsequent amendments. The Act regulates the conditions under which alcohol may be bought and sold and the minimum purchase age. It also sets out penalties for breaching those conditions.
- Advertising of alcohol is controlled by the Broadcasting Act (1993), and restrictions on driving after drinking alcohol are dealt with in the Land Transport Act (1998) and its amendments. The Resource Management Act (1991) and the Local Government Act (2001) also have provisions that allow local government bodies to determine where alcohol is bought and sold in their district.
- The Alcohol Law Reform Bill currently before Parliament aims to reduce the access to alcohol by limiting the range of outlets that may sell alcohol, controlling the density of alcohol outlets, reducing the alcohol content of some drinks and tightening up on marketing. The Bill also proposes raising the purchase age for off licensed premises and higher penalties for some liquor-related offences.
- Most New Zealanders drink at least some alcohol, and many drink more than the recommended amount. Recent surveys that have examined the drinking habits of New Zealanders have found that young people (including underage drinkers) are particularly likely to drink excessively. The surveys also established that there was a general tolerance of drunkenness among those surveyed.
- The effects of alcohol on health are well documented. Alcohol impairs cognitive and motor function, decreases inhibitions and so increases risk-taking behaviour, accidents and other injuries. Exposure to alcohol in the critical developmental period before birth and again during adolescence is particularly damaging, potentially causing life-long behavioural and cognitive consequences.
- Certain individuals and families are particularly susceptible to problems with alcohol. While genetic components are known to be responsible, at least in part, their interaction with easy access to alcohol and low social constraints are also likely to play a part, along with factors such as stress, adverse life experiences and gender roles.
- Major categories of disease are causally linked to heavy alcohol drinking including neuropsychiatric, gastrointestinal, and cardiovascular conditions and a range of cancers. Alcohol is a contributing or exacerbating factor for a wide range of other conditions such as depression, diabetes, sleep disorders, infectious diseases and inflammatory conditions, and injuries from burns, falls and fractures.
- Although light to moderate drinking has been associated with a decreased risk of coronary disease in healthy older adults, heavy drinking as well as occasional binge drinking at any age has the opposite effect. Moreover, alcohol is contraindicated for many other conditions, including other cardiovascular diseases.

- Intoxication with alcohol is the main cause of alcohol-related harm in New Zealand society, leading to risk-taking behaviour, road traffic and other accidents, injuries from falls and drowning, assaults, self-inflicted injury, and acute alcohol poisoning.
- There are social costs from the use and abuse of alcohol, to drinkers themselves, to others they come into contact with, and to the whole of New Zealand society. Costs range from avoidable morbidity and mortality, hospital presentations, aggression and violence, and crime, to harmful effects on family relationships, friendships and social life, schooling, study and employment opportunities, and financial problems.
- Alcohol-related harm is not evenly distributed in the New Zealand population. New Zealanders with low socioeconomic status suffer disproportionately compared to those who are better off. Similarly, the burden of alcohol-related harm varies by ethnicity and is higher for the Māori and Pacific populations.
- Children whose caregivers abuse alcohol are more likely to suffer from abuse and neglect, and to be exposed to family violence. They also have an increased risk of injury from vehicle and other accidents and non-accidental injury. Additionally, increased exposure to alcohol at an early age makes it more likely that children will have alcohol problems themselves later on.
- Alcohol-related harm has a significant economic impact including the cost of injuries, adverse effects on work and study productivity, vandalism, and loss of trade because of public safety concerns.
- Approaches that combine reducing the availability of alcohol and modifying the drinking environment along with restrictions on advertising and sponsorship and increased enforcement of drink-driving counter-measures have been shown to be successful in reducing alcohol-related harm. Community-wide approaches to discourage heavy drinking and encourage alcohol-free events and entertainment also have a positive influence.
- The Position Paper on Alcohol for which this background paper provides the supporting evidence is a proposed joint statement by the five South Island District Health Boards, indicating their concern about the extent of alcohol-related harm in New Zealand and the need for action to reduce it.

## **Table of Contents**

Prepared by:	1
Willow MacDonald, Public Health South, Southern DHB	1
Sarah Colhoun and Susan Bidwell, Community and Public Health, Canterbury DF	IB 1
Executive summary	2
Executive summary	2
Table of Contents	4
Section One: Introduction and overview of alcohol in New Zealand	5
Availability of alcohol	6
Current legislation	7
Driving	8
Advertising	9
Alcohol Reform Bill	9
NZ drinking pattern	10
Section Two: The effects of alcohol on health	11
Alcohol and the brain	11
Genetic and environmental influences on drinking behaviour	11
Effect of alcohol on foetal development	12
Effect of alcohol on adolescents	12
Alcohol and older people	13
Alcohol and disease	14
Cancers	14
Diseases of the liver and pancreas	15
Cardiovascular disease	16
Depression	16
Infectious diseases	16
Type 2 diabetes	17
Section Three: Alcohol-related harm	17
Mortality	17
Hospital presentations	18
Traffic crashes	18
Crime	19
Social costs	20
Children and young people	21
Māori	22
Economic costs and benefits	22
Section Four: Reducing alcohol-related harm	23
Increasing taxation	23
Reducing the purchase age	24
Reducing availability	24
Modifying the drinking context	25
Restricting marketing, promotion and sponsorship	26
Drink-driving counter-measures	26
Acting to reduce alcohol-related harm	26
References	27

# Section One: Introduction and overview of alcohol in New Zealand

This background paper has been developed to provide supporting evidence to the accompanying Position Statement on Alcohol. It is set out in four sections: Section One describes how alcohol is used in New Zealand and gives an overview of existing and proposed legislation. Section Two covers the personal health effects of alcohol in terms of its effect on the brain and its contribution to a range of diseases. Section Three outlines the social and economic harms caused by alcohol. Section Four presents the approaches that have been shown to be effective in reducing alcohol-related harm.

The paper was written collaboratively by staff from Community & Public Health and Public Health South. It was internally peer reviewed and then circulated for feedback to a wide range of people in the South Island DHB networks and beyond.

Searches of the international databases Medline and Science Citation Index were conducted in October 2011 to gather evidence for the effects of alcohol on all aspects of health and disease, and the social harm created by alcohol.

A wide ranging search of other electronic information was undertaken to identify information on the status of alcohol in New Zealand and to locate international and New Zealand examples of measures taken or suggested to reduce alcohol related harm. Major sources included:

- Policy and position statements from New Zealand District Health Boards
- The Canterbury District Health Board submission to the Law Commission review of all aspects of the law on the sale and supply of liquor
- Publications from the Ministry of Health, Statistics New Zealand, Ministry of Transport, the New Zealand Police, and the Ministry of Social Development.
- The report of Law Commission *Alcohol in Our Lives: Curbing the Harm* (Law Commission 2010)
- University research groups, particularly the University of Otago Injury Prevention Research Unit
- The Alcoholic Liquor Advisory Council
- Alcohol Health Watch.
- Considerable use was also made of the two editions of the work by Babor et al (2003 & 2010) in the publication *Alcohol: No Ordinary Commodity. Research and Public Policy.*

References to other relevant sources that were cited in the works investigated were also followed up for further information.

Alcoholic beverages contain the substance ethanol (pure alcohol), which is produced when yeast ferments carbohydrates or sugars in food (ALAC 2011). Alcoholic beverages usually contain between 2.5% and 55% alcohol, ranging from light beer to brandy, rum and whisky. The latest alcoholic products available are ready-to-drink beverages (RTDs), also known as alcopops. These have been available in New Zealand since 1995, and usually range in alcohol content from 4 to 7%. In New Zealand, beverages need to contain at least 1.15% alcohol to be defined as liquor (or

alcohol). The Sale of Liquor Act (1989) defines liquor as "any fermented, distilled, or spirituous liquor (including spirits, wine, ale, beer, porter, honeymead, stout, cider, and perry) that is found on analysis to contain 1.15% or more alcohol by volume".

Alcoholic beverages are measured in terms of 'standard drinks', defined as 10 grams of pure alcohol. Beverages with the highest pure alcohol content require the least volume to comprise one standard drink. Spirits with 40% alcohol content in a 30ml glass, wine with 13% alcohol content in a 100 ml glass, RTDs with 5% alcohol content in a 275ml bottle, or beer with an alcohol content of 4% in a 330 ml can are all examples of standard drinks.

### Availability of alcohol

Alcohol is widely available in New Zealand. Statistics New Zealand figures (2011) show that in 2010, 474 million litres of alcoholic beverage were available for consumption, of which the pure alcohol content was 33 million litres (Statistics New Zealand 2011). This meant that on average, each New Zealander aged 15 years and over had available 9.6 litres of pure alcohol to consume.<sup>1</sup> In the same year the amount of pure alcohol available increased 5.5% from the previous year with the sharpest increase being a 20% increase in the amount of spirits. The amount of wine available also increased by 7.7%. The most widely available alcoholic beverage in 2010 by volume was beer (299 million litres available), followed by wine (103 million litres) and spirits (13 million litres).

New Zealanders surveyed for a Ministry of Health report on alcohol use in 2007/08 (MoH 2009) most commonly consumed wine (62.3% of those aged 16-64 years) and beer (60.6%). A large number also reported drinking spirits, liqueurs or mixed cocktails (46.5%), while a smaller number said they consumed RTDs (22.2%), and fewer still drank sherry, port or vermouth (6.7%) or cider (5.4%).

Currently in New Zealand, anyone who is over the age of 18 may purchase alcohol to drink. Those who are under 18 years of age may drink on licensed premises only if the alcohol has been purchased by a legal parent or guardian, and if they remain in a supervised or unrestricted area of the premises. In order to purchase alcohol, people may need to supply photographic identification as proof of age. In New Zealand there is no minimum drinking age (Wellington City Council undated).

New Zealanders can purchase alcohol from a variety of locations, defined by the type of licences which apply to the sale of alcohol. 'On-licences', including bars, restaurants, hotels, and cafes, sell alcohol on the premises for consumption on the premises. 'Off-licences' sell alcohol for consumption off the premises and include bottle-stores, supermarkets, and convenience stores and dairies. Club licences enable the operators to sell alcohol for consumption on the premises to members of the club or to their invited guests. Special licences allow alcohol to be sold at special events. (Christchurch City Council 2011).

The number of places to drink in New Zealand has increased since 1989 when the Sale of Liquor Act lifted restrictions on liquor licensing, meaning more and more places have been able to sell

<sup>&</sup>lt;sup>1</sup> Figures on the amount actually consumed were sought during the writing of this paper but do not appear to be readily available.

alcohol. In 1990, the number of places which held liquor licences was 6295; by 2010 this number was 14, 424 (Law Commission 2010).

The density of alcohol outlets has been found to be associated with alcohol-related harm in both international and national research. A New Zealand study by Connor et al. (2011) noted that "more than 50 research papers have been published since the early 1990s finding associations between the spatial density of alcohol outlets and levels of harm" (p. 841). The same study showed that the number of licensed premises within 1km of home was associated with the level of self-reported harm and that the number of off-licences was associated with binge drinking.

As well as the wide variety and number of locations where alcohol is available, New Zealanders can purchase alcohol at any time and almost all days of the year (ALAC undated). Since 1989, 24 hour liquor licences have become available, meaning that bars, supermarkets and convenience stores may supply alcohol throughout the day and night. Since 1999 (with an amendment to the Sale of Liquor Act), New Zealanders have been able to purchase alcohol on Sundays. In 2004, hours of sale were extended to include Easter Sunday for wineries. On public holidays the sale of alcohol may be restricted, depending on the type of premises. Pubs and off-licences may not sell alcohol on Christmas Day, Good Friday, Easter Sunday or before 1pm on Anzac Day. (ALAC undated)

Alcohol is available at relatively cheap prices in New Zealand. There are no minimum prices for alcohol – price is determined by the industry. The average price of alcohol has increased over the last 10 years, but by comparison, so has the price of milk. The average prices for alcohol in 2010 were as follows: 1 litre whisky \$42.94; 3 litre cask white wine \$21.80; 1 dozen bottles beer \$19.50; and glass of beer (licensed premises) \$4.95 (Gunesekara and Wilson 2010). These prices represent increases of 18.8%, 26.5%, 16.2% and 35.1% respectively over the ten year period since 1999. In comparison, the price of milk has increased by 22.8%. Although the price of alcohol has increased, it has become more affordable, in comparison with average hourly earnings. For example, results of this study compare differences in "the minutes taken to earn sufficient alcohol to reach the legal blood alcohol limit" from 1999 to 2010. In 1999 it would have taken a worker 16.4 minutes to earn enough money to get intoxicated by drinking whisky, while in 2009 it would only have taken 13.2 minutes. Prices of alcohol can be subject to wide variation; the same study found that 3 litres of cask wine could be bought for as little as \$16.99 (Gunesekara and Wilson 2010). A more recent study (Sloane et al. 2011) compared discounted alcohol prices to that of discounted non-alcoholic beverages. The average lowest price per standard drink was: \$0.61 for cask wine; \$1.17 for cider; 0.69 for RTDs; \$0.33 for Coca Cola; \$0.35 for apple juice; \$1.39 for grape juice; and \$0.42 for milk. The difference in price between cask wine and grape juice, both made from the same product, was noted by the authors.

Alcohol prices in New Zealand are subject to excise tax (ie a tax on specific goods sold or produced for sale, paid by the producer or seller to the government). Alcohol is taxed either by its ethanol content within a range or by beverage volume. For example, beverages with an alcohol content of between 9-14% are taxed at 10%. In 2009, tax rates were 10% of the retail price for beer, 15% for wine, and 38% for spirits (Alcohol Healthwatch undated).

### **Current legislation**

New Zealand's current main piece of legislation regarding alcohol is the Sale of Liquor Act (1989). Under this Act, the sale of alcohol is able to be determined by the market, rather than by 'need' (Maclennan 2010, p.48). Alcohol sales are no longer monopolised by a few providers, but are

driven by "free-market ideology" (Maclennan 2010, p.48). The Sale of Liquor Act lifted restrictions on hours and places of sale, allowing 24 hour licenses and supermarkets to sell wine. The Act aimed to allow communities greater control of liquor licensing, with most of the responsibility for liquor licensing going to local government. It also introduced the concept of host responsibility, which means, for example, that licensed premises are required to have food and non-alcoholic beverages available to purchase. The Act aimed to "promote moderate drinking and reduce alcohol-related harm through host responsibility" (Maclennan 2010, p.48). The Act had several amendments in 1999 which lowered the purchase age from 20 to 18 years; allowed licensed premises to sell alcohol on Sundays; and enabled supermarkets and grocery stores to sell beer as well as wine.

Effectively, the Sale of Liquor Act made it easier to sell alcohol, by lifting restrictions. Its aims were "... to establish a reasonable system of control over the sale and supply of liquor to the public with the aim of contributing to the reduction of liquor abuse, so far as that can be achieved by legislative means [and] to set out the provisions relating to on-license, off-licenses, club licenses and special licenses; define the powers of the Licensing Authority, District Licensing Agencies; outline offences and enforcement provisions" (National Drug Policy New Zealand 2010). The Sale of Liquor Act introduced District Licensing Authorities (DLAs), in each local area, with powers to grant and renew liquor licenses in their area and stipulate hours of trade (Maclennan 2010, p.60). Each DLA has licensing inspectors who monitor how the licensed premises are run (for example, not selling alcohol to those under the age of 18 or serving alcohol to intoxicated customers).

New Zealand law imposes penalties on those who breach the regulations around the sale of alcohol. Those who try to purchase alcohol before they are legally of age (i.e. under 18 years) can receive a \$200 fine. If they are unsupervised on licensed premises they may incur a \$2000 fine, and if they are apprehended drinking or possessing alcohol in a public place they may be fined \$300. Those who try to use someone else's ID to purchase alcohol may be charged with fraud and receive heavy penalties if convicted (New Zealand Drug Foundation undated).

There are also penalties for those who attempt to supply alcohol to minors (under 18 year-olds). Those who purchase alcohol for supply to under 18 year-olds can receive a fine of up to \$2000. On licensed premises, managers and staff can be fined heavily (separate fines for each) for selling or supplying alcohol to those who are underage

(a fine of up to \$10,000 for the manager, up to \$2000 for a staff member, and a suspended licence for the licensee). For the sale or supply of alcohol to an intoxicated person the licensee may be fined up to \$10,000.

Apart from the Sale of Liquor Act (1989), there are two other pieces of legislation relevant to the way alcohol is supplied and controlled. Under the Resource Management Act (1991), local government bodies have the power to determine the location of new licensed premises, and under the Local Government Act (2001) they have the power to introduce bylaws and enforce them with penalties of up to \$20,000.

### Driving

In New Zealand the law sets the limits for the amount of alcohol a person can drink while driving measured by Blood Alcohol Content (BAC). Drivers under 20 years of age have a zero BAC limit (lowered in 2011 from 0.03 milligrams per 100 millilitres of blood). This means that after even one drink a young person can be charged with drink driving. Fines and demerit points are the penalties

for a blood alcohol level of less than 150 micrograms per litre of breath and less than 30 milligrams per 100 millilitres of blood. For higher blood alcohol content, disqualification, demerit points, fines and imprisonment are all available as penalties (New Zealand Transport Agency 2011). Penalties for drivers 20 years and over who attempt to drive with a BAC in excess of the permitted levels include up to three months in prison, and/or a fine of up to \$4500, and loss of licence for six months or more for a first or second offence. Penalties increase for repeat offenders. For those who kill or injure someone while driving under the influence of alcohol, penalties are higher - up to five years' imprisonment and/or a fine of up to \$20,000.

### Advertising

Currently legislation around the advertising of alcohol is controlled by the Broadcasting Act (1993), which allows for the self-regulation of alcohol advertising by the alcohol industry. Within the advertising industry itself, the Advertising Standards Authority is responsible for alcohol advertising codes (Maclennan 2010, p.49). Prior to the 1980s no advertising of alcohol at all was allowed through any media. By 1981, off-licenses were allowed to advertise on television and radio. In 1987, alcohol companies were allowed to sponsor sports and advertise corporately. In 1992 advertising of alcohol products was allowed on broadcast media between 9pm-6am on television and anytime on radio (Maclennan 2010, p.49).

### Alcohol Reform Bill

In 2008 the Government commissioned the Law Commission to review legislation around alcohol in New Zealand. After an extensive review, the Law Commission produced the report, "Alcohol in our Lives, Curbing the Harm" (Law Commission 2010). The Government responded to this with the Alcohol Law Reform Bill.

The Alcohol Law Reform Bill was first introduced in November 2010. It has had two readings so far and is currently (as at November 2011) before Parliament. It has received more than 5000 submissions. The then Minister of Justice, Simon Power, stated that the Bill called for a "safe and responsible drinking culture" and aimed to "reduce excessive drinking, improve the operation of the alcohol licensing system (including community input on licensing), support the responsible sale, supply and safe consumption of alcohol...and make licenses harder to get and easier to lose".<sup>2</sup>

The Alcohol Reform Bill responds to the Law Commission report by picking up

"126 of the 153 recommendations in the Law Commission report" (Nick Smith, Second Reading, Parliament 2011). According to the Minister of Justice, for the

"first time in 20 years Parliament aims to restrict, rather than relax, our liquor laws" (Simon Power, Second Reading, 2011).

The Bill aims to reduce the accessibility of alcohol by limiting the maximum national trading hours: 8am-4am for on-licenses, clubs and special licenses; 7am-11pm for off-licenses. It also targets off-licenses, including convenience stores and dairies, which would no longer be able to sell alcohol.

<sup>&</sup>lt;sup>2</sup> See <u>http://www.parliament.nz/en-NZ/PB/Legislation/Bills/8/2/7/00DBHOH\_BILL10439\_1-Alcohol-Reform-Bill.htm</u> for documents and media releases relating to this Bill and from which the statements in this and the following two paragraphs are taken.

Supermarkets would only be able to display alcohol in one non-prominent area. The Bill would instigate a split purchase age: the off-license purchase age would increase to 20, while on-licenses would stay at 18. The Bill would also make the supply of alcohol to minors without parental or guardian consent an offence, and would put greater emphasis on host responsibility. It aims to give communities greater control about the use of alcohol in their own area, giving them the responsibility to make decisions about the location of alcohol outlets in their area, along with controlling the density of alcohol outlets and opening hours. It would aim to limit the alcohol content of RTDs to 5%, and by law "particularly dangerous alcohol products would be able to be banned." The new law would also create tighter limits on alcohol marketing. There would be a higher penalty for breaching the liquor ban and an investigation of minimum prices. While these measures go some way towards implementing the recommendations of the Law Commission report, they have not been considered strong enough by many. Kypri et al (2011), for example, described them as "tinkering" rather than reform.

### NZ drinking pattern

Most New Zealanders drink at least some alcohol. Eight in ten of the New Zealanders aged between 16-64 (85.2%) who were surveyed in the 2007/08 New Zealand Alcohol and Drug Use Survey had had an alcoholic drink in the past year (MoH 2009). In the same survey, nearly two-thirds (61%) drank alcohol at least once a week. A smaller proportion (7%) drank alcohol daily – equating to 152, 900 adults in the New Zealand population. Men were more likely than women to drink more often, with 6.3% of men drinking daily compared to 4.3% of women (MoH 2009).

In New Zealand, the current legal purchase age for alcohol is 18 years, however, surveys show that many New Zealanders start drinking at a younger age. In 2007/08, for those surveyed aged between 16-64, the median age for first trying alcohol was 16 years. Almost one third of New Zealanders (31.9%) had first tried alcohol at age 14 or younger (MoH 2009).

Many New Zealanders also drink more than the recommended amount per drinking occasion (more than 6 standard drinks for males and more than 4 standard drinks for females). In 2007/08, nearly two-thirds (61.6%) of drinkers exceeded the recommended amount at least once during the year. This is most common for young drinkers: for those in the 18-24 year age-group, 80% had consumed more than the recommended amount at least once in the last year (MoH 2009). Almost half of those surveyed (44.3%) had first consumed a large amount of alcohol when they were aged between 15 and 17 years; 13.9% had consumed a large amount of alcohol at least once when aged 14 years or younger. Maori and Pacific groups were also more likely to drink large amounts: 76.6% of Maori and 76.5% of Pacific drinkers had exceeded the recommended amount at least once in the last year. Almost one in ten (9.4%) New Zealanders who drank more than the recommended allarge amount did so one or two times a week in the past year. In addition, 3.2% consumed a large amount three or more times a week. The majority of New Zealanders surveyed (76.7%) had consumed a large amount of alcohol at least once).

Another nationally representative survey that sampled 628 young people aged between 12 and 17 years and 659 adults aged 18 years and over found that New Zealanders tolerate drunkenness (ALAC 2005). A quarter (27%) of those surveyed agreed that "It's OK to get drunk as long as it's not every day". One third of people (34%) disagreed that "It's never OK to get drunk" while less than half of people (47%) agreed with this statement. Nearly one in ten drinkers (9%) "drink to get drunk". Drunkenness was more prevalent amongst young people. Nearly half (47%) of those between 12 and 17 years agreed that "It's OK to get drunk as long as it's not every day". Within the same age group, 14% said they "drink to get drunk". The survey had a margin of error of

 $\pm 4.3\%$  for the youth sample and  $\pm 5.5\%$  for the adult sample (both at the 95% confidence level) and results were weighted in terms of ethnicity, age and gender.<sup>3</sup>

### Section Two: The effects of alcohol on health

The hazardous and harmful use of alcohol contributes to death, disease and injury to the drinker and to others through the dangerous actions of intoxicated people, or the impact of drinking on foetal and child development. The most recent World Health Organisation report on alcohol (2011), categorised it as the world's third largest risk factor for disease and disability, a causal factor in 60 types of disease and injuries and a component cause in 200 others. The WHO report noted that almost 4% of all deaths worldwide can be attributed to alcohol, more than the deaths caused by HIV/AIDS, violence or tuberculosis. New Zealand researchers have suggested that ethanol shares the same characteristics as other substances that are classified as drugs of high risk to public health and it should be scheduled accordingly (Sellman et al 2009). This section provides an overview of the health effects of alcohol drawn from international journal literature, including recent meta-analyses.

### Alcohol and the brain

Alcohol acts on specific targets in the brain and triggers a variety of neurotransmitter systems that lead to acute behavioural effects (Field et al. 2010). Alcohol impairs cognitive and psychomotor function, affects mood, and causes feelings of intoxication, and impairments in memory, attention and planning. It decreases inhibitions and so increases sexual risk taking, aggressive behaviour and risk of motor vehicle accidents and other injuries.

The prefrontal cortex, the hippocampus, the cerebellum, the white matter and glial cells of the brain are particularly susceptible to the effects of alcohol. Alcohol also acts on neurotransmitters, and signalling pathways resulting in neural impairments and cognitive and behavioural dysfunction. Some mechanisms are common to both the developing and the adult brain, while others affect developmental stages before birth and during adolescence.

Mechanisms and signalling pathways that affect reinforcement and reward are involved in the initiation and maintenance of alcohol drinking behaviour. The biological mechanisms that link impulsivity and inhibitory control over alcohol are as yet poorly understood (Field et al. 2010; Lejuez et al. 2010). The impairment of control results in diminished ability to focus on and persist with tasks, a tendency to act on the spur of the moment, poor future planning, inability to delay gratification, and diminished ability to regulate emotion (Lejuez et al. 2010). Additionally, the impairment of inhibitory control is associated with increased alcohol seeking behaviour.

### Genetic and environmental influences on drinking behaviour

<sup>&</sup>lt;sup>3</sup> The methodology of this study is described

The linkage and association of alcohol dependence with particular chromosomes and genes has been the subject of much research. The US National Institute on Alcohol Abuse and Alcoholism, for example, has been funding the Collaborative Studies on Genetics of Alcoholism (COGA) since 1989. The goal of these studies is to identify the specific genes which underlie the vulnerability of some individuals to having problems with alcohol.<sup>4</sup> However, while the initiation and maintenance of problem drinking behaviour can be attributed at least in part to genetic mechanisms, environmental factors are also important, and the two interact in a complex way (Campbell & Oei 2010; Young-Wolff et al. 2011). Twin and adoption studies where genetic and environmental factors can be controlled have established that there is a variance in interfamilial alcohol problems; certain children do not develop alcohol use disorders in spite of having a family history of them, while others with no family history may do so (Campbell & Oei 2010). A wide range of environmental factors that may interact with a genetic predisposition have been investigated. Social learning cognitions that form expectancies and memories around alcohol in childhood are believed to play a critical part by some researchers (Oei & Morawska 2004), while other studies have looked at factors such as gender roles, stress, adverse life experiences, wider societal influence, and the age of drinking onset (Harden et al. 2008; Pitkanen et al. 2008; Winstanley et al. 2008; Latendresse et al. 2010). Low social constraints and easy access to alcohol are believed to be particularly influential at younger ages (Kendler et al. 2011).

### Effect of alcohol on foetal development

The developing brain is particularly vulnerable to alcohol. Alcohol affects the developing central nervous system, leading to a range of physical, learning, and behavioural deficits including growth deficiencies, cranio-facial abnormalities, and intellectual disabilities (Alfonso-Loeches & Guerri 2011). Heavy pre-natal exposure to alcohol has been associated with deficits in memory, language, attention, learning, visuo-spatial abilities, fine and gross motor skills, and social and adaptive functioning. Exposed children are more likely to be rated as hyperactive, disruptive, impulsive or delinquent than non-exposed children and have difficulties with social mixing and aggression, anxiety disorders, and suicidal ideation throughout their lives (Hellemans et al. 2010).

MRI studies in children affected by prenatal alcohol exposure have found structural alterations to the shape, volume and surface area of the overall brain and particular brain regions, as well as reduced white matter and increased grey matter densities in corresponding areas (Guerri et al. 2009). The specific brain structures affected are strongly influenced by the timing of alcohol exposure, with the most damaging period being early in pregnancy and up until week 20. Binge drinking appears to have a more damaging effect than chronic exposure to lower levels of alcohol (Alfonso-Loeches & Guerri 2011). The genetic make up of the mother and foetus also has an influence. If either or both have the aldehyde dehydrogenase enzyme, which provides more efficient alcohol metabolism at higher blood concentration levels, there is some measure of protection (McCarver et al. 1997).

### Effect of alcohol on adolescents

The adolescent brain is still maturing, undergoing important structural and functional changes particularly in the pre-frontal cortex (Witt 2010). It is particularly vulnerable to alcohol toxicity, alcohol and substance use problems and psychiatric disorders. The area most affected by alcohol is

<sup>&</sup>lt;sup>4</sup> See <u>http://pubs.niaaa.nih.gov/publications/arh26-3/214-218.htm</u>

that associated with cognitive flexibility, self-regulation, inhibitory control, and judgement of risk and reward. These brain circuits develop relatively late in adolescence and may explain the tendency of adolescents to impulsiveness and to disregard the consequences of their behaviour, both of which also increase the risk of substance abuse. Heavy drinking at this age can have a negative impact on brain structure and functions that cause short and long term cognitive and behavioural consequences (Maldonado-Devincci et al. 2010; Alfonso Loeches & Guerri 2011; Witt 2010). MRI studies of adolescents with alcohol use disorders have established that they had smaller brain volumes in the prefrontal cortex compared with control subjects (De Bellis et al. 2005). There also appear to be gender differences in how vulnerable the different areas of the adolescent brain are to alcohol, with females being more at risk than males, although the mechanisms whereby this occurs are not yet fully understood (Caldwell et al. 2005; Guerri & Pascual 2010; Alfonso Loeches & Guerri 2011).

Adolescent binge drinking is associated with cognitive deficits, poor academic achievement, and attention and memory disorders. Adolescent drinkers are susceptible to the immediate consequences of alcohol use, including blackouts, hangovers, and alcohol poisoning and are at elevated risk of neurodegeneration (particularly in regions of the brain responsible for learning and memory), impairments in functional brain activity, and neurocognitive deficits. Drinking in early adolescence may modify brain maturation processes in certain areas that result in a worsened ability in problem solving, verbal and non-verbal retrieval, visuo-spatial skills, and working memory. Heavy episodic or binge drinking impairs study habits and erodes the development of transitional skills to adulthood (Ziegler et al. 2005).

Another important long-lasting consequence of alcohol use during adolescence is the higher risk of developing alcohol abuse and dependence in adulthood. Adolescents who begin to use alcohol before the age of fifteen are four times more likely to develop alcohol dependence at some stage of their lives compared with those who start at the age of twenty or later (De Wit et al. 2000). It is not clear whether starting to drink at an early age has a causative relationship with alcoholism or whether it suggests a predisposition because of personality characteristics, genetic background, and environmental factors (Alfonso Loeches & Guerri 2011; Kendler et al. 2011).

### Alcohol and older people

While there is evidence for some benefits of moderate consumption of alcohol in healthy adults, there is considerable risk associated with even moderate intake in older people. Alcohol in older age has a wide range of negative associations (Heuberger 2009). Physiological changes with ageing mean that older people have poorer balance, are frailer and more at risk of falls. They are more likely to be taking prescribed or over-the-counter medications that interact with alcohol, or to have existing diseases which contraindicate drinking alcohol. Older people often have poor appetites and alcohol displaces important nutrients which would help to maintain health. In conjunction with ageing and poor diet, alcohol has also been associated with dental and oral disorders such as impaired salivary flow, periodontitis, and poor chewing function, which further deplete nutritional status. Older people are at greater risk for gastrointestinal disease and cancer, and are more likely to suffer from bereavement, to be depressed, or socially isolated, all of which may precipitate heavy drinking. Moreover, "excessive intake" in an older person may be equivalent to an amount that would be considered moderate in someone younger. However, wide differences in individual medical status and life history modify the extent that these factors apply to any particular individual.

Longitudinal studies have suggested that light to moderate alcohol consumption may have a protective effect against age-related pre-dementia, Alzheimer's disease or vascular origin dementia, though findings from research have been inconsistent (Panza et al. 2009). On the other hand, there is no evidence to show that light drinking is a risk factor for these conditions, but neither is there any evidence of a specific beneficial level.

### Alcohol and disease

The ethanol in alcoholic drinks is oxidised in the liver to acetaldehyde, and further detoxified to acetate. Variants in genetic make up result in differences in the ability of individuals to process acetaldehyde, leading to a varied genetic susceptibility to alcohol exposure. Major categories of disease are causally linked to alcohol, particularly heavy drinking. These diseases include neuropsychiatric, gastrointestinal, and cardiovascular conditions, a range of cancers, foetal alcohol syndrome and pre-birth complications (WHO 2011). Alcohol is also a contributory cause to a wide range of other conditions including major depression, anxiety, diabetes, sleep disorders, infectious diseases such as pneumonia and tuberculosis, and injuries from burns, falls, and fractures (Carrao et al. 2004). Among conditions which are exacerbated by alcohol are inflammatory conditions such as psoriasis (Farkas & Kemeny 2010) and there is a relationship between heavy drinking and epileptic seizures (Samokhvalov et al. 2010). Alcohol, especially beer, increases the risk of gout (Singh et al. 2011) and contributes to the risk of obesity as it is a passive form of energy consumption adding to the energy consumed as food (Yeomans 2010). Chronic heavy drinking can also irreversibly compromise bone quality and increase the risk of osteoporosis in later life (Sampson 2002).

The harmful use of alcohol is a particularly important risk factor for men. The World Health Organisation (2011) classes alcohol as the leading risk factor for death in males ages 15–59, mainly due to injuries, violence and cardiovascular diseases. Globally, 6.2% of all male deaths are attributable to alcohol, compared to 1.1% of female deaths. Men also have far greater rates of total disease burden attributed to alcohol than women – 7.4% for men compared to 1.4% for women. Men outnumber women four to one in weekly episodes of heavy drinking and have much lower rates of abstinence. Lower socioeconomic status and educational levels are also associated with a greater risk of alcohol-related death, disease and injury – and are social determinants that are greater for men (WHO 2011).

### Cancers

Alcohol consumption is strongly associated with cancers of the head and neck, (the oral cavity, pharynx, larynx, and oesophagus) as well as the respiratory and digestive tract. Daily consumption of around 50g of alcohol increases the risk for these cancers by two to three times, compared with the risk in non-drinkers. Acetaldehyde produced from alcohol by microbes in the normal gastrointestinal tract or salivary glands is a local and topical carcinogen that affects the human gut (Salaspuro 2003; Chen et al. 2009; Oze et al. 2011; Ogden 2005). The relationship is dose dependent, so that damage increases with heavier drinking. There is uncertainty about the existence of a threshold (a number of drinks per day) below which alcohol may not cause any damage but it is likely that any threshold varies depending on ethnicity, gender, age, and the existence of other conditions. Some ethnic groups, for example Japanese, have a genetic inability to detoxify acetaldehyde (aldehyde dehydrogenase deficiency). These populations have a greatly increased risk

of alcohol-related GI tract cancers compared to populations where most people have a fully active enzyme (Baan et al. 2007; Salaspuro 2003).

Alcohol and tobacco have a synergistic effect greater than either of them alone so that heavy smokers and drinkers have a greatly increased risk of upper aero-digestive tract cancers. Acetaldehyde is also present in high concentrations in cigarette smoke, hence the increased exposure to acetaldehyde from both smoking and drinking (Ansary-Moghaddam et al. 2009; Altieri et al. 2005; Ogden 2005). It has been suggested that this exponentially greater effect may be due to the direct contact of solvent action on the tissues of the body that enhances the effects of tobacco (Altieri et al. 2005).

There is good evidence from several systematic reviews that heavy alcohol intake moderately increases the risk of breast cancer (Weir et al. 2007; Baan et al. 2007; Collaborative Group on Hormonal Factors in Breast Cancer 2002). One systematic review (Carrao et al. 2004) observed a 7.1% increase in the risk of breast cancer for each additional 10g per day intake of alcohol compared to non-drinkers. The estimated risk ratio at 25g alcohol per day was 1.25, at 50g/day was 1.55 and at 100g per day was 2.41.

Similar results have been reported from the pooled analysis of studies of the effect of alcohol on colorectal cancer, with evidence of an increased relative risk of about 1.4 for colorectal cancer with regular consumption of 50g alcohol per day compared to non-drinkers (Baan et al. 2007).

### **Diseases of the liver and pancreas**

Liver cirrhosis is characterised by the replacement of normal tissue with fibrous tissue and the loss of functional liver cells. It leads to permanent scarring of the liver that blocks blood flow and prevents normal metabolic and regulatory processes. The causal impact of heavy drinking on liver disease and on cirrhosis in particular has been known for several hundred years, and has been confirmed by recent studies (Corrao et al. 1999; Corrao et al. 2004). A meta-analysis by Rehm et al. (2010) that further investigated the dose-response relationship found that more than two drinks a day for women and more than three drinks a day for men were significantly associated with higher risk, and the risk increased with increasing volumes of alcohol consumed. This meta-analysis also suggested that for people with any signs of liver disease (even when not related to alcohol), only abstinence from alcohol could be considered safe. Moreover, there were likely to be interactions between alcohol and drugs that are used to treat liver problems (Rehm et al. 2010, p. 442).

There is a link between alcohol consumption and pancreatitis – both in its acute and chronic forms. Alcohol has numerous deleterious effects on the pancreas including direct toxicity to pancreatic cells as well as changes of production and flow of pancreatic juice which result in mechanical obstruction of pancreatic ducts, fibrosis, and toxicity to other organ systems from the by-products of ethanol metabolism. A meta-analysis (Irving et al. 2009) of studies that had addressed the relationship between alcohol and pancreatitis found that there was evidence supporting a threshold effect (at four drinks daily) and an exponential dose-response relationship above that threshold between average volume of alcohol consumed and the risk of pancreatitis.

### Cardiovascular disease

There is a complex relationship between cardiovascular disease and drinking of alcohol. Moderate consumption (one drink/15g alcohol per day for women or two drinks/30g alcohol for men) has been associated with a decreased risk for coronary heart disease in observational studies of diverse populations (Rimm et al. 1999). Brien et al. (2011) confirmed this protective effect based on a meta-analysis of interventional studies<sup>5</sup> that measured circulating blood levels of biomarkers linked to the risk of coronary heart disease. They concluded that there was "…compelling indirect evidence in support of a causal protective effect of alcohol" (p. 13).

This protective effect on coronary disease is often referred to as the "J-shaped curve" whereby moderate consumption of alcohol in healthy people is beneficial but excessive or binge drinking is quite the opposite (Costanzo et al. 2010). However, even occasional excessive consumption appears to wipe out any protective effect on the heart even if overall consumption is generally moderate or light (Roerecke & Rehm 2010). In contrast, alcohol consumption has detrimental effects on other cardiovascular diseases regardless of drinking pattern (WHO 2011). It is associated with sudden cardiac death, although the effect is complex and the mechanisms are not completely understood. For people with cardiomyopathy or arrhythmias, avoiding alcohol altogether is the most favourable option for reducing risk (George & Figueredo 2010). Alcohol also increases blood pressure, and increases morbidity and mortality from stroke independently from smoking (WHO 2011). Binge drinking is a significant risk factor for stroke especially in people with hypertension (Hillborn et al. 2011). Alcohol is also not recommended for young people, pregnant women, those at risk of alcoholism, or anyone whose activity calls for concentration, skill or coordination (Roerecke & Rehm 2010).

### Depression

A meta-analysis of the associations between alcohol use disorder and major depression (Boden & Fergusson 2011) found that the presence of either disorder doubled the risk of the other. The association could not be accounted for fully by common factors that influence both conditions and the disorders appeared to be linked in a causal manner, with the most plausible association being that alcohol use disorder increases the risk of major depression, rather than vice versa. Potential mechanisms include neurophysical and metabolic changes linked to exposure to alcohol but further research is needed to clarify the exact nature of the linkage. Alcohol is also associated with suicidality in both adolescents and adults (O'Connell & Lawlor 2005; Galaif et al. 2007).

### Infectious diseases

A recent meta-analysis (Samokhvalov et al. 2010) showed a strong and consistent relationship between alcohol and the risk of community acquired pneumonia. A relative risk of 8.22(95% CI 4.85-13.95) was found after adjusting for age, sex, smoking and other confounding factors such as comorbidities. Additionally, there was a clear dose-response relationship with increasing consumption giving a greater risk. Mechanisms involved include detrimental immunological effects of alcohol, and diminished oropharyngeal tone resulting in increased risk of aspiration and

<sup>&</sup>lt;sup>5</sup> The studies included in this meta-analysis administered controlled amounts of alcohol to participants and measured the effect on various biomarkers (for example circulating blood levels of HDL cholesterol, fibrinogen etc), compared to a period of no use of alcohol.

suppression of coughing. Heavy drinking also predisposes people to liver damage, nutritional deficiency, and poor personal hygiene which may result in immunity being further impaired. The same factors are likely to be involved in the elevated risk of tuberculosis in heavy drinkers (Lonnroth et al. 2008).

### Type 2 diabetes

A dual relationship exists between alcohol consumption and type 2 diabetes. Light to moderate drinking may be beneficial while heavy drinking is detrimental. A recent meta-analysis (Baliunas et al. 2009) found that there was a U shaped relationship between average alcohol consumption and risk of diabetes in both men and women, although the biological mechanism for this is not clear. However, the risk of alcohol consumption on other diseases and health outcomes at even moderate levels may outweigh any positive benefits of alcohol consumption for diabetes.

### Section Three: Alcohol-related harm

Alcohol use causes approximately 4% of deaths worldwide and 4.5% of the global burden of disease. This is not only because of the deleterious effect of alcohol on the drinker's health but because of the social problems caused by alcohol to the drinker and others. This places alcohol beside tobacco as a leading preventable cause of death and disability (World Health Organisation, 2011; Babor et al., 2010). Intoxication is the main cause of alcohol-related harm in New Zealand society 'because it can lead to risk-taking behaviour, accidents and injuries, violence, and acute alcohol poisoning' (World Health Organisation, 2007). Binge drinking sessions in particular are associated with acute health consequences including: road traffic injuries, falls, drowning, poisoning, assault, self-inflicted injury, and foetal alcohol spectrum disorder (Kypri et al. 2003). This section of the background paper addresses the wide range of alcohol-related harm experienced in New Zealand, including mortality, hospital presentations, crime, and social and economic costs. The ways that alcohol-related harm negatively impacts on children, Māori and people with low socioeconomic status are also explored.

### Mortality

Research by Connor et al. (2004) estimated that 3.9% of all deaths in New Zealand in 2000 were attributable to alcohol consumption (approximately 1040 deaths). This study calculated that some deaths from coronary disease would have been prevented by light to moderate drinking of alcohol. However, these were almost entirely among older people, while those attributed to alcohol occurred before middle age with injury being a major contributor. The authors estimated that 17,200 years of life were lost but only 5,300 years of life gained, a net loss of almost 12,000 years of life due to alcohol in one year in New Zealand (Connor et al 2004, p.3) The study also noted that alcohol-related mortality is experienced unequally throughout the population, with men having four to five times the rate of alcohol attributable years of life lost compared to women. Māori men and women were also found to have higher mortality and rates of years of life lost than non-Māori of the same age (Connor et al. 2004).

### **Hospital presentations**

Research on presentations to an Auckland Emergency Department indicated that 35% of injurybased emergency department presentations were alcohol-related (Humphrey et al. 2003). The New Zealand Alcohol and Drug Use Survey found that 4.7% of adults (aged 16-64 years) had experienced an injury in the past year due to their alcohol use. Men (7.2%) were found to be significantly more likely than women (5.1%) to have experienced an injury in the past year due to their alcohol use (Ministry of Health, 2009).

Data collected from Dunedin Hospital Emergency Department for the period 1 November 2010 to 29 October 2011, identified 892 patients who were seen with alcohol-related presentations. This is considered an underestimate and does not include the victims of others' drinking. Of these 892 cases, 65% (582) were seen on a Friday, Saturday or Sunday. Over half the cases, 55% (492), were seen between 9pm and 6am, a time when staffing and other hospital resources are at their lowest. The length of stay in the Emergency Department was over four hours for 62% (560) of these presentations. The average length of stay for uncomplicated intoxicated patients was 4.5 hours with an average cost to the tax payer of \$1000.

Alcohol-related presentations to Emergency Departments throughout New Zealand impact negatively upon staff. A recent study of Wellington Regional Hospital Emergency Department found that a large proportion of *all* Emergency Department staff regularly experienced aggressive and abusive behaviour from intoxicated patients. A large proportion of Emergency Department nurses and ambulance staff also reported experiencing physical assault from intoxicated patients (Gunasekara, et al. 2011). Intoxicated patients presenting to Emergency Departments were reported to negatively impact the quality of care for other patients, for example other patients frequently reported feeling threatened, which could contribute to the stress of their conditions (Gunasekara et al., 2011).

An investigation of the wholly alcohol-attributable hospital admissions for the period 1986 to 2006 indicates the majority of these admissions were for mental and behavioural disorders due to alcohol use. Admissions for mental and behavioural disorders were highest in the 35 to 44 year age group, but occurred in relatively large numbers for all age groups. The largest relative and absolute increase in wholly alcohol-attributable hospital admissions was in the 15 to 19 years age group, which increased by 126% over this period. This is arguably a result of the lowering of the purchase age of alcohol in 1999 (Law Commission 2010).

### Traffic crashes

It is well established that alcohol consumption degrades driving performance and negatively impacts driving behaviour. Numerous studies also illustrate that the risk of being involved in a vehicle accident increases as a driver's blood alcohol level increases (Ministry of Transport, 2010). Driver consumption of alcohol causes significant harm in New Zealand through death, injury, and the associated social costs (Ministry of Transport, 2010).

In 2009 driver alcohol and or drug use was a contributing factor in 113 fatal traffic crashes, 420 serious injury crashes and 1,107 minor injury crashes. These crashes resulted in 138 deaths, 576 serious injuries and 1,743 minor injuries. The total social cost of crashes

involving alcohol or drugs was about \$890 million; that is nearly a quarter of the social cost associated with all injury crashes. (Ministry of Transport, 2010).

Twenty seven percent of drivers in all fatal crashes between 2007 and 2009 were reported as having consumed alcohol. Alcohol and drugs were a contributing factor in almost 21% of all serious injury crashes and 13% of minor injury crashes in New Zealand between 2007 and 2009 (Ministry of Transport, 2010).

The harm associated with alcohol-related traffic crashes is often experienced by those other than the alcohol impaired driver, for instance 40% of alcohol-related traffic crash injury is experienced by those other than the alcohol impaired driver (Connor & Casswell, 2009).

For every 100 alcohol or drug-impaired drivers or riders killed in road crashes, 56 of their passengers and 26 sober road users die with them (Ministry of Transport, 2010).

### Crime

Alcohol intoxication is associated with violence and aggression:

...a causal link between alcohol intoxication and aggression has been supported by epidemiological and experimental research, as well as by research indicating specific biological mechanisms linking alcohol to aggressive behaviour. Experimental studies suggest a causal relationship between alcohol and aggression...although this relationship is clearly moderated by gender and personality as well as by situational and cultural factors (Babor et al., 2010, p.46).

New Zealand Police's 2009 *National Alcohol Assessment* clearly demonstrates the significant contribution of alcohol to crime in New Zealand. Of all recorded offences in the year 2007/08:

- At least 31% of offences were committed where the offender had consumed alcohol prior to committing the offence
- At least one third of violence offences occurred where the offender had consumed alcohol prior to committing the offence

The number of alleged offenders identified as having consumed alcohol prior to offending increased from 27% in 2005/06 to 32% in 2007/08 (New Zealand Police, 2009).

Alcohol consumption has been identified as a factor in homicides in New Zealand. Either a suspect or victim was found to have been under the influence of alcohol in 49.5% of the 489 homicides recorded between 1999 and 2008 (New Zealand Police, 2009). A Ministry of Social Development investigation of 141 family violence-related homicides found that alcohol and or drug abuse was a factor in approximately two thirds of the couple-related homicides, and was common as both a factor in perpetrator's backgrounds and as a factor at the time of the event (Ministry of Social Development, 2009).

Alcohol and or drug abuse also featured strongly in child homicides, causing researchers to conclude that:

...children are at highest risk of death from maltreatment in their first year of life and when they live with young unemployed parents or caregivers who abuse alcohol and drugs (Ministry of Social Development, 2009).

District Court Judges have also commented on the impact of alcohol use and abuse on the justice system, estimating that up to 80% of defendants coming before the criminal courts have alcohol or other drug use abuse or dependency issues, and that alcohol is the drug of choice in three-quarters of these cases (Law Commission, 2009).

### Social costs

New Zealanders experience a range of social harms due to their own, or other people's alcohol use and abuse. The New Zealand 2007/08 Alcohol and Drug Use Survey found that one in eight adults<sup>6</sup> (12.2%) had experienced harmful effects from their own drinking in the past year. These included harmful effects on friendships or social life, home life, work/study/employment opportunities, financial position, and legal problems or difficulty learning (Ministry of Health, 2009). These harms were not distributed evenly throughout the population. Men (9.8%) were more likely than women (6.1%) to be affected, and men living in the highest deprivation area (NZDep2006 quintile 5) (12.9%) being more likely than men living in the least deprived area (NZDep2006 quintile 1) (5.3%) to report that they had experienced harmful effects on friendships and social life due to their own drinking (Ministry of Health, 2009).

New Zealanders with lower socioeconomic status bear a disproportionate burden of alcohol-related harm. People living in more socioeconomically deprived neighbourhoods are generally more likely than people living in less deprived areas to have experienced :

- harmful effects in the last year due to their own alcohol use
- harmful effects on their home life in the past year due to someone else's alcohol use
- assault (physical and/or sexual) in the past 12 months due to someone else's use of alcohol or drugs (Ministry of Health, 2009).

Research also shows that many New Zealanders experience harm caused by someone else's drinking. In the 2007/8 New Zealand Alcohol and Drug Use Survey one in six adults (18.1%) had experienced harmful effects on their friendships/social life, home life or financial position due to someone else's drinking. Significantly more women (22.8%) than men (17.0%) had experienced such harm from someone else's drinking (Ministry of Health, 2009).

Recent New Zealand research (Casswell et al., 2011a) found a significant association between a person's exposure to a heavy drinker and reduced personal wellbeing and poorer health status. Over 28% of the study sample reported having one or more heavy drinkers in their lives. This research indicates that a large proportion of New Zealanders report harm from the drinking of others. This includes a wide range of harms, such as being physically hurt, emotionally hurt and neglected, and lower work productivity. A large proportion reported at least one adverse event that was attributed to the drinking of a stranger (Casswell et al., 2011b). An Australian national study that surveyed a randomly selected sample of 2,649 adults (Laslett et al) found that 70% of them reported experiencing nuisance, fear, or abuse from strangers' drinking and 30% reported negative effects from someone close to them, whether family, friends, or co-workers.

<sup>&</sup>lt;sup>6</sup> Included adults aged 16-64 (Ministry of Health, 2009).

Heavy drinking makes a significant contribution to workplace absenteeism, impaired work performance, lack of productivity, and low morale both when consumed within or outside of normal work hours (Bacharach et al 2010; Spicer et al 2003; Frone 2006; Bery et al 2007). A study of 2,805 employed adults in the United States, for example, found that 15% reported consuming alcohol before work, during the work day, working under the influence, or working with a hangover (Frone 2006). Data from an Australian national survey showed that high risk drinking occurred at least occasionally in 44% of workers (Bery et al 2007). Other studies have found drinking, hangovers, and a workplace culture that condones high alcohol use are significant contributors to arguments with supervisors and co-workers, falling asleep on the job, harassment of female employees, and medical problems and injuries (Ames et al 1997; Bacharach et al 2007; Bery et al 2007).

### Children and young people

2009).

There are numerous concerns about the impact of alcohol on children and young people's wellbeing. Aside from the effects of alcohol on foetal development already discussed above, children may suffer from:

- increased susceptibility to child abuse and neglect if caregivers have an alcohol problem
- increased likelihood of exposure to family violence
- increased risk of injury from other persons under the influence of alcohol, ie. vehicle accidents, accidental and non-accidental injury, and
- increased risk of early exposure to alcohol themselves (Law Commission, 2010, Girling et al., 2006).

Children are particularly vulnerable to harms caused by others' drinking. Research about people with a heavy drinker in their lives found that 17% of respondents with children under 18 years old indicated their children had been negatively affected by someone else's drinking in the last 12 months. This included being verbally abused and witnessing violence (Casswell et al., 2011b).

As earlier stated, caregivers with alcohol problems featured strongly in a review of child homicides in New Zealand (Ministry of Social Development, 2009). Another review, looking at the deaths of children between 2005 to 2007<sup>7</sup> found that alcohol was a significant contributing factor in a range of child deaths. Alcohol was found to be involved in 31% of the 158 deaths of children due to vehicle crash injuries.<sup>8</sup> Alcohol was also involved in 19.1% of the 199 child deaths due to drowning, assault, poisoning, suffocation or falls (Child and Youth Mortality Review Committee, 2009).

Infants and younger children (under 15 years of age) had significantly fewer deaths involving alcohol (2.2% of 92 deaths) compared to young people (15 to 24 years of age) (32.1% of 265 deaths) (Child and Youth Mortality Review Committee, 2009).

Alcohol contributes to child deaths differently, depending on age. For infants and young children, alcohol impacts the quality of care giving, contributes to family violence and leads to undersupervision of vulnerable infants and children. In contrast there is a dramatic increase in death rates

<sup>&</sup>lt;sup>7</sup> This review focussed on motor vehicle deaths during 2007 and deaths other than motor vehicle (specifically drowning, assault, suffocation, poisoning and falls) from 2005 to 2007 (Child and Youth Mortality Review Committee, 2009). <sup>8</sup> This included 16 as drivers, 17 as passengers and 14 as pedestrians (Child and Youth Mortality Review Committee,

from injury for those 15 years and older, which is largely related to adolescent risk-taking behaviour which alcohol consumption contributes to (Child and Youth Mortality Review Committee, 2009).

Early exposure to alcohol (consumption on multiple occasions before 15 years of age) is linked to a range of poor adult outcomes, including substance dependence, criminal conviction, herpes infections and failure to achieve educational qualifications (Law Commission, 2010, p.89).

The negative impact of alcohol on adolescents was also identified by the New Zealand Alcohol and Drug Use Survey, which found that a quarter (23.0%) of people aged 16 to 17 years had experienced harmful effects on their friendships or social life in the past year due to someone else's drinking, and one in five (18.8%) had experienced injuries in the past year due to their own alcohol use (Ministry of Health, 2009).

Foetal exposure to alcohol during gestation can cause Foetal Alcohol Syndrome (FAS) or Foetal Alcohol Spectrum Disorder (FASD), which in turn, are associated with a range of adverse life experiences, including disrupted schooling experiences, trouble with the law, confinement (in jail, prison, or a psychiatric or alcohol/drug inpatient settings), inappropriate sexual behaviours, and drug and alcohol problems (Streissguth, et al., 2004).

### Māori

Evidence clearly demonstrates that Māori suffer disproportionately from a wide range of alcoholrelated harms compared to non-Māori. This is shown by:

- Māori being more likely to die from alcohol-related causes (Māori have 4.2 times the alcohol related mortality of non-Māori and double the rate of years of life lost due to alcohol) (Connor et al., 2005)
- Māori men being significantly more likely to experience harmful effects on financial position, work, study or employment, experience injuries or having legal problems due to alcohol use (Ministry of Health, 2009)
- Māori women being twice as likely to experience harmful effects on home life, social life or financial position, experience injuries or legal problems due to alcohol use (Ministry of Health, 2009)
- Māori women being four times more likely than non-Māori women to have been assaulted by someone who had consumed alcohol (Ministry of Health, 2009)
- Māori being more likely to be apprehended by police for an offence that involved alcohol (New Zealand Police, 2009).

### Economic costs and benefits

Alcohol-related harm in New Zealand has a significant economic cost, including the cost of injury, adverse effects on businesses and employment issues. Harmful alcohol use in 2005/06 alone cost New Zealand an estimated \$4,794 million of diverted resources and lost welfare. Injuries from harmful alcohol use in 2005/06 cost an estimated \$1.592 billion in New Zealand (Slack et al., 2009).

Businesses have reported various day-to-day adverse effects of alcohol including vandalism, environmental hazards that need to be cleaned up (eg. urine, vomit, broken glass) and loss of trade due to public safety concerns (Law Commission, 2010).

Alcohol has a negative impact on work and study productivity, with the Alcohol and Drug Use Survey finding that:

- 3.2% of adults reported harmful effects to their work, study or employment opportunities as a result of alcohol use in the last year
- An estimated 5.6% of respondents reported having one day off school or work in the last 12 months as a result of alcohol use
- Nearly one in ten adults worked while feeling under the influence of alcohol (Ministry of Health, 2009).

Alcohol does have a number of economic benefits to society in terms of creating employment and export earnings. According to the National Business Review (Swann 2009), in 2009 the wine industry alone was worth \$1.5 billion to New Zealand's gross domestic product, and was the 11<sup>th</sup> largest export, representing 2.2% of total goods exported. The same article noted that the wine industry provided 16,568 full time jobs and contributed to tourism, with over 200,000 visitors annually to vineyards and related events. These tourists also on average stayed longer and spent more than other tourists. The export of spirits and liqueurs has also increased in recent years, with a contribution of \$52.6 million to the economy in the year 2010-2011 (National Business Review 2011).

### **Section Four: Reducing alcohol-related harm**

As outlined in the previous sections, there is strong evidence that New Zealanders' lives are affected by alcohol in numerous ways ranging from preventable mortality, and morbidity, to the effects of violence and lost productivity. It has been argued that this harm has resulted from an increasingly liberalised environment for the sale of alcohol, and societal attitudes which encourage excessive drinking. This section looks at measures that have been shown to be effective in reducing alcohol-related harm. Much of it draws on the work by Babor et al (2010) *Alcohol: no ordinary commodity* which documents the reasons why effective alcohol policies need to take account of the physical, social and economic environments in which people live rather than merely concentrating on the health effects alone.

### **Increasing taxation**

Increasing alcohol taxes has been shown to be highly effective in reducing alcohol consumption and therefore alcohol related harm. Moreover it is not expensive to implement. Evidence shows that when prices go up, alcohol consumption goes down (Babor et al 2010; Ministry of Justice 2007).

Currently, New Zealand's excise tax rate is lower than other nations. The tax on beer in New Zealand for example (10%), is much lower than that in Australia (24%) (Casswell and Maxwell 2005). New Zealand also has an excise tax policy that groups alcoholic beverages into 'bands' – for instance, alcoholic beverages with an alcohol content between 9-14 % are taxed at the rate of 10% (Alcohol Healthwatch undated). This means that manufacturers have an incentive to create higher-alcohol content drinks in order to get the best tax advantage. Instead, the excise tax could be based directly on the level of ethanol in the product.

Alcohol prices are subject to the economic theory of supply and demand. That is, when there is an increased supply and high demand for alcohol, or when there is a constant supply but decreased demand, prices go down. Thus, "deliberate changes in prices will affect the supply and demand relationship" (Babor et al. 2010, p.110). Therefore a strategy for government is to increase excise taxes on alcohol (and increase revenue), leading to increased alcohol prices which in turn leads to decreased alcohol consumption. Other factors affecting alcohol prices (apart from excise tax) include differences in price between on-premise and off-premise outlets and the geographical distance and transport needed to access supplies of alcohol.

There has been a worldwide trend towards reduced alcohol prices since 1950. One reason for this is that excise taxes are set at a fixed amount, and are not adjusted for inflation. Unless legislation sets a new tax level, in line with inflation, excise taxes on alcohol do not reflect the value of the local currency.

Alcohol price changes may particularly affect youth and heavy drinkers. Studies have shown that young people drink less when beer prices go up (Coate and Grossman 1988). Babor et al. (2010, p. 121) draw attention to a number of studies that have shown higher beer taxes have significantly reduced both the frequency of youth drinking and the probability of heavy drinking, with heavy drinkers being more affected by price increases than occasional drinkers (Laixuthai and Chaloupka 1993). These studies also noted a reduction in violence in association with higher alcohol prices. Higher prices may also result in changes in health outcomes. In the U.S. between 1960 and 1975, when liquor taxes (and prices) went up, cirrhosis mortality went down (Cook 1981).

### Reducing the purchase age

Changes to the minimum purchase age have been shown to substantially affect youth drinking. For example, in the 1970s and 1980s in the United States, the minimum alcohol purchase age fluctuated greatly. One study showed that when the minimum purchase age was increased to 21 it led to reduced alcohol use among young Americans and a decrease in traffic crashes (O'Malley and Wagenaar 1991). Other studies in the U.S. have shown that the effectiveness of limiting supply to youth is dependent upon enforcement – otherwise young people are still able to access alcohol (Wagenaar and Wolfson 1994).

### **Reducing availability**

Reducing the physical availability of alcohol is a strategy that is effective yet relatively inexpensive to implement. The cost of restricting alcohol availability is low relative to the health consequences of drinking. The main methods of reducing the availability of alcohol are through restricting hours and days of sales, and controlling the number, location, and type of retail premises. Regulations can be imposed such as government ownership of alcohol outlets, outlet location and limiting 'bunching' (e.g. minimum distance between outlets), limiting the number of outlets, restricting hours and days of sale, and restricting the density of retail outlets (Babor et al. 2010). Studies have shown that changing the hours and days of sale has "significant impacts on the volume of alcohol consumed and the rates of alcohol- related problems" (Babor et al. 2010, p.136). Restricting sales of alcohol from off-licence premises is most likely to affect those who do not keep a supply of alcohol.

Another factor influencing the availability of alcohol is the location and density of liquor outlets. Governments can impose laws limiting the location of alcohol outlets, and these can be enforced at local and national level. For example, zoning laws may impose restrictions on the location of outlets nears schools and churches. Density of outlets can be regulated by declaring a minimum distance between them, or by limiting the number of outlets in an area, or both. The 'bunching' of bars and restaurants, is considered to give rise to alcohol-related problems. Traffic crashes in particular are more likely to occur in high-density alcohol outlet areas. Changing the number of outlets in an area can therefore influence consumption of alcohol and alcohol-related problems (Babor et al. 2010, p.131).

Further restrictions can be put in place concerning how alcohol is sold at on-premise and offpremise locations. There is likely to be more control over sales from on-premises than offpremises. With on-premise sales there is room to influence a variety of factors around the sale of alcohol including specifying drink sizes, non-discounted drink promotions, and responsible and trained service.

Government ownership of alcohol outlets has been shown to reduce the number of outlets, limit the hours of sale, and remove the profit motive. There is evidence that off-premise monopoly systems limit alcohol consumption and alcohol-related problems; and conversely, that the elimination of government off-premise monopolies can increase alcohol consumption. For example, Swedish alcohol consumption increased when beer became available in grocery stores between 1965 and 1977.

It is possible to further restrict the physical availability of alcohol through regulating who may sell alcohol. Those who sell alcohol are required to hold a license to do so. It is common to check the credentials of those who seek licenses, for example, excluding those with a criminal history. The Law Commission report recommended tightening up regulations on who may obtain a liquor licence – they must be a "suitable person" and declared that there were "serious problems" with laws relating to off-licences, particularly the number of small liquor outlets (p.14). The report recommended prohibiting service stations and takeaway food stores from selling alcohol and that spirits and RTDs should only be available from specialist alcohol outlets (Law Commission Report Summary, p.14).

### Modifying the drinking context

Efforts to reduce alcohol harm can be focused on the environments where alcohol is bought and consumed. It has been shown that aspects of the bar environment such as serving practices that promote intoxication, aggressive bar staff and the inability to manage problem behaviour make alcohol-related problems worse (Babor et al. 2010, p.149). Approaches to modifying the drinking context include responsible beverage service training and in-house policies for licensed premises, enhanced enforcement of sanctions for breaching the regulations relating to the sales of alcohol, the legal liability of servers, managers and owners of licensed premises, voluntary codes of practice, interventions for managing aggression and other problem behaviour, reducing environmental precipitants of aggression, and community mobilization approaches (Babor et al. 2010, p.149-162). Limiting the strength of alcoholic beverages; and promotion of alcohol-free activities are also likely to have an influence.

### Restricting marketing, promotion and sponsorship

Advertising of alcohol has increased in many countries over recent decades, and studies have shown it to have detrimental effects. Increasingly, countries have left the regulation of alcohol promotion to the self-regulation of the industry. Research shows that alcohol advertising can have a detrimental effect on young people, exposing them early to the messages that alcohol is attractive, glamorous and fun. If these messages permeate the environment at a young age they may set drinking patterns in later life. Effectively, alcohol advertising leads to greater consumption of alcohol (Babor et al. 2010, p.188-189). A further effect of alcohol advertising is to colour how people perceive the drinking habits of others, thus normalising drinking behaviour. Since the early 1990s, sponsorship of sports and cultural events has become common, particularly those that appeal to young people. These events combine marketing messages on products, giveaways, and grounds signage and associate them with good times and with drinking the sponsors product. McCreanor et al (2008), draw attention to the Export Gold campaign in 2004 that used as its slogan "The best weekend you'll never remember" in a competition aimed at young people, describing it as "...the synergistic, cumulative effects of environmental exposure of young people to alcohol marketing [which] creates and maintains expectations and norms for practices of drinking to intoxication." (p. 944)

Legislation may attempt to partially restrict alcohol advertising, for example by banning the advertising of spirits, or restricting hours of advertising to particular times on television. Research into the effect of such legislation has been inconclusive to date, largely because of methodological difficulties and limited experience internationally of countries that have implemented comprehensive restrictions. Babor et al (2010), however, concluded that the "...most probable scenario, based on the theoretical and empirical evidence available, is the extensive restriction of marketing would have an impact" (p. 188).

### Drink-driving counter-measures

Alcohol greatly impairs how well a person can drive. As levels of alcohol in the blood go up, driving performance declines. Research has shown that judgement and reaction times are impaired at a BAB of 0.05% and substantially diminished at 0.10% (Davis et al 2003 cited in Babor et al 2010, p. 166). Impairment risk goes up exponentially with increasing amounts of alcohol: at a BAC of 0.08% the relative risk of any crash involvement is 2.7 and at a BAC of 0.15% the relative risk is 22.1 (Babor et al 2010, p. 166). Various measures can be instigated to counter this, one of which is to set maximum Blood Alcohol Concentration (BAC) levels (see Section One). Enforcement includes random breath-testing to check these levels, and penalties for those who exceed the drink-driving BAC limits. Further restrictions may also be placed on younger or inexperienced drivers, for example, reducing the legal BAC level to 0 for drivers under 20, as has just been done in New Zealand in 2011.

### Acting to reduce alcohol-related harm

There is increasing concern about the misuse of alcohol in New Zealand society and the need to take action to reduce the harm it causes.

In their report on curbing the harm from alcohol, the Law Commission Report (2009) proposed an 'Alcohol Harm Reduction Act' which would:

... encourage responsible attitudes to the promotion, sale, supply and consumption of alcohol; contribute to the minimisation of crime, disorder and other social harms; delay the onset of young people drinking alcohol; protect and improve public health; promote public safety and reduce public nuisance; and reduce the impact of the harmful use of alcohol on the Police and public health resources.

In order to reduce alcohol-related harm, the Commission recommended:

...restricting the times alcohol can be sold; restricting the places alcohol can be sold; preventing a growing proliferation of alcohol outlets; increasing the purchase age for alcohol; expanding the grounds upon which a liquor licence can be declined; providing for more local input into liquor licensing decisions; reorganising and upgrading the efforts of local authorities in relation to alcohol decisions; providing for local alcohol policies to be decided by councils; financing liquor licensing, monitoring and compliance through licence fees, not a combination of fees and ratepayer contributions; tightening the law about off-licences; and improving regulation of special licences" (Law Commission Report Summary, p.13).

The South Island District Health Boards are gravely concerned about the effect that misuse of alcohol has, either directly or indirectly, on the health and wellbeing of all New Zealanders and particularly the disproportionate effect that it has on the most vulnerable members of society. In their role as leaders responsible for promoting and protecting the health of their constituents the Boards are publishing a joint Position Statement which indicates that concern and sets out a number of recommendations for actions that can and should be taken to address the harm created by alcohol.

### References

Ames, G.M., Grube, J.W., Moore, R.S. 1997. The relationship of drinking and hangovers to workplace problems: an empirical study. Journal of Studies on Alcohol 58(1), 37-47.

ALAC (Alcoholic Liquor Advisory Council). Undated. Strengthening community action on alcohol. Available: <u>http://www.alac.org.nz/research-resources/pdfs-alac-resources</u> Accessed 30.11.11.

ALAC (Alcoholic Liquor Advisory Council). Undated. Drinking and driving. Available: <u>http://www.alac.org.nz/alcohol-you/drinking-driving</u> Accessed 30.11.11.

ALAC (Alcoholic Liquor Advisory Council). 2005. The way we drink 2005 – Executive Summary. ALAC Occasional Publication No. 27. Wellington: ALAC.

ALAC (Alcoholic Liquor Advisory Council). 2011. What's in a standard drink. Available: <u>http://www.alac.org.nz/alcohol-you/whats-standard-drink</u> Accessed 29.11.11

Alcohol Health Watch. Undated. Information sheet: tax on alcohol. Available: <u>http://www.ahw.org.nz/resources/Fact%20Sheet/Info%20Sheet%20Alcohol%20and%20Tax%20Fi</u>nal%2009.pdf Accessed 30.11.11.

Alcohol Health Watch.2004.Briefing paper.Alcohol excise tax: changes to the New Zealandsystem.Auckland:AlcoholHealthWatch.Available:http://www.ahw.org.nz/resources/pdf/Excisetax221dec2004.pdfAccessed 30.11.11.

Alfonso-Loeches, S., Guerri, C. 2011. Molecular and behavioral aspects of the actions of alcohol on the adult and developing brain. Critical Reviews in Clinical Laboratory Sciences 48(1), 19-47.

Altieri, A., Garavello, W., Bosetti, C., Gallus, S., La Vecchia, C. 2005. Alcohol consumption and risk of laryngeal cancer. Oral Oncology 41(1), 956-965.

Ansary-Moghaddam, A., Huxley, R.R., Lam, T.H., Woodward, M. 2009. The risk of upper aero digestive tract cancer associated with smoking, with and without concurrent alcohol consumption. Mount Sinai Journal of Medicine 76(4), 392-403.

Baan, R., Straif, K., Grosse, Y., Secretan, B., El Ghissassi, F., Bouvard, V. et al. 2007. Carcinogenicity of alcoholic beverages. Lancet Oncology 8(4), 292–293, 2007.

Bacharach, S.B., Bamberger, P., Biron, M. 2010. Alcohol consumption and workplace absenteeism: the moderating effect of social support. Journal of Applied Psychology 95(2), 334-348.

Bacharach, S.B., Bamberger, P.A., McKinnery, V.M. 2007. Harassing under the influence: the influence of male heavy drinking, the embeddedness of permissive workplace drinking norms, and the gender harassment of female coworkers. Journal of Occupational Health Psychology 12(3), 232-250.

Baliunas D.O., Taylor, B.J., Irving, H., Roerecke, M., Patra, J., Mohapatra, S., Rehm, J. 2009. Alcohol as a risk factor for type 2 diabetes: a systematic review and meta-analysis. Diabetes Care, 32(11), 2123–2132.

Babor, T., Caetano, R., Casswell, S., Edwards, G, Giesbrecht, N., Graham, K., Grube, J. et al. 2010. *Alcohol: no ordinary commodity. Research and public policy.* 2<sup>nd</sup> ed. New York: Oxford University Press.

Baliunas D.O., Taylor, B.J., Irving, H., Roerecke, M., Patra, J., Mohapatra, S., Rehm, J. 2009. Alcohol as a risk factor for type 2 diabetes: a systematic review and meta-analysis. Diabetes Care, 32(11), 2123–2132.

Bery, J.G., Pidd, K., Roche, A.M., Harrison, J.E. 2007. Prevalence and patterns of alcohol use in the Australian workforce: findings from the 2001 National Drug Strategy Household Survey. Addiction 102, (9), 1399-1410.

Boden, J.M., Fergusson, D.M. 2011. Alcohol and depression. Addiction 106(5), 906-914.

Brien, S.E., Ronksley, P.E., Rurner, B.J., Mukamal, K.J., Ghali, W.A. 2011. Effect of alcohol consumption on biological markers associated with risk of coronary heart disease: systematic review and meta-analysis of interventional studies. BMJ 342, d636. doi: 10.1136/bmj.d636.

Caldwell, L.C., Schweinsburg, A.D., Nagel, B.J., Bartlett, V.C., Brown, S.A., Tapert, S.F. 2005. Gender and adolescent alcohol use disorders on BOLD (blood oxygen level dependent) response to spatial working memory. Alcohol and Alcoholism 40(3), 194-200.

Campbell, J.M., Oei, T.P. 2010. A cognitive model for the intergenerational transference of alcohol use behavior. Addictive Behaviors 35(2), 73-83.

Casswell, S., Maxwell, A. 2005. What works to reduce alcohol-related harm and why aren't the policies more popular? Social Policy Journal of New Zealand: Te Puna Whakaaro Issue 25 (July).

Casswell, S., You, R.Q., and Huckle, T. 2011a. Alcohol's harm to others: reduced wellbeing and health status for those with heavy drinkers in their lives. Addiction 106 (6), 1087-1094.

Casswell, S., Harding, J.F., You, R.Q., and Huckle, T. 2011b. Alcohol's harm to others: self reports from a representative sample of New Zealanders. New Zealand Medical Journal 124(1336):75-84.

CDC (Centers for Disease Control). 2008. Alcohol and public health: frequently asked questions. Available: <u>http://www.cdc.gov/alcohol/faqs.htm#whatAlcohol</u> Accessed 29.11.11.

Chen, L., Gallicchio, L., Boyd-Lindsley, K., Tao, X.G., Robinson, K.A., Lam, T.K. et al. 2009. Alcohol consumption and the risk nasopharyngeal carcinoma: a systematic review. Nutrition and Cancer 61(1), 1-15.

Christchurch City Council. 2011. Types of liquor licences. Available <u>http://www.ccc.govt.nz/business/licencesconsents/liquorlicences/typesoflicences.aspx</u> Accessed 29.11.11.

Child and Youth Mortality Review Committee, Te Rōpū Arotake Auau Mate o te Hunga Tamariki. 2009. *Special Report: The involvement of alcohol consumption in the deaths of children and young people in New Zealand during the years 2005-2007.* Wellington: Child and Youth Mortality Review Committee.

Coate D., Grossman M. 1988. Effects of alcoholic beverage prices and legal drinking ages on youth alcohol use. Journal of Law and Economics 31(1), 145-71.

Collaborative Group on Hormonal Factors in Breast Cancer. 2002. Alcohol, tobacco and breast cancer: collaborative reanalysis of individual data from 53 epidemiological studies, including 58,515 women with breast cancer and 95,067 women without the disease. British Journal of Cancer, 87(11), 1234-1245.

Connor, J., Broad, J., Jackson, R., Vander Hoorn, S., Rehm, J. 2004. *The burden of death, disease and disability due to alcohol in New Zealand*. Wellington: ALAC Available: <u>http://www.alac.org.nz/sites/default/files/research-publications/pdfs/BurdenExec.pdf</u> Accessed 29.11.11.

Connor, J., Casswell, S. 2009. The burden of road trauma due to other people's drinking. Accident Analysis & Prevention 41(5): 1099-1103.

Connor, J.L., Kypri, K., Bell, M.L., Cousins, K. 2011. Alcohol outlet density, levels of drinking and alcohol-related harm in New Zealand: a national study. Journal of Epidemiology and Community Health 65(10), 841-846.

Cook, P.J. 1981. The effect of liquor taxes on drinking, cirrhosis and auto fatalities. In: Moore M., Gerstein D. (eds.) *Alcohol and public policy: beyond the shadow of prohibition*, p.255-85. Washington, DC: National Academy of Sciences.

Corrao, G., Bagnardi, V., Zambon, A., & La Vecchia, C. 2004. A meta-analysis of alcohol consumption and the risk of 15 diseases. Preventive Medicine 38(5), 613-619.

Corrao G., Bagnardi, V., Zambon, A., Arico, S. 1999. Exploring the dose-response relationship between alcohol consumption and the risk of several alcohol related conditions: a meta-analysis. Addiction 94(10), 1551-1573.

Corrao G., Bagnardi, V., Zambon, A., Vecchia, C.L. 2004. A meta-analysis of alcohol consumption and the risk of 15 diseases. Preventive Medicine 38(5), 613-619.

Costanzo, S., Castelnuovo, A.D., Donati, M.B., Iacoviello, L., de Gaetano, G. 2010. Cardiovascular and overall mortality risk in relation to alcohol consumption in patients with cardiovascular disease. Circulation 121(17), 1951-1959.

De Bellis, M.D., Narasimhan, A., Thatcher, D.I., Keshavan, M.S., Soloff, P., Clark, D.B. 2005. Alcoholism Clinical and Experimental Research 29(9), 1590-1600.

De Wit, D.J., Adlaf, E.M., Offord, D.R., Ogborne, A.C. 2000. Age at first alcohol use: a risk factor for the development of alcohol disorders. American Journal of Psychiatry 157(5), 745-750.

Farkas, A., Kemeny, L. 2010. Psoriasis and alcohol: is cutaneous ethanol one of the missing links? British Journal of Dermatology 162(4), 711-716.

Field, M., Wiers, R.W., Christiansen, P., Fillmore, M.T., Verster, J.C. 2010. Acute alcohol effects on inhibitory control and implicit cognition: implications for loss of control over drinking. Alcoholism: Clinical and Experimental Research 34(8), 1346-1352.

Frone, M.R. 2006. Prevalence and distribution of alcohol use and impairment in the workplace: a U.S. National Survey. Journal of Studies on Alcohol 67(1), 147-156.

Galaif, E., Sussman, S., Newcomb, M.D., Locke, T.F. 2007. Suicidality, depression, and alcohol use among adolescents: a review of empirical findings. International Journal of Adolescent Medicine & Health 19(1), 27-35.

George, A., Figueredo, V.M. 2010. Alcohol and arrythmias: a comprehensive review. Journal of Cardiovascular Medicine 11(4), 221-228.

Girling, M., Huakau, J., Casswell, S., Conway, K. 2006. *Families and Heavy Drinking: Impacts on children's wellbeing –systematic review*. Blue Skies Report No. 6/06. Wellington: Families Commission. Available: <u>http://www.nzfamilies.org.nz/sites/default/files/downloads/BS-families-and-heavy-drinking.pdf</u> Accessed 30.11.11.

Guerri C., Bazinet, Q., Riley, E.P. 2009. Foetal alcohol spectrum disorders and alterations in brain and behaviour. Alcohol and Alcoholism 44(2),108-114.

Guerri, C., Pascual, M. 2010. Mechanisms involved in the neurotoxic, cognitive, and neurobehavioral effects of alcohol consumption during adolescence. Alcohol 44(1), 15-26.

Gunasekara, F., Butler, S., Cech, T, Curtis, E. Douglas, M., Emmerson, L. et al. 2011. How do intoxicated patients impact staff in the emergency department? An exploratory study. New Zealand Medical Journal 124(1336), 14-23.

Gunasekara, F., Wilson, N. 2010. Very cheap drinking in New Zealand: some alcohol is more affordable than bottled water and nearly as cheap as milk. New Zealand Medical Journal 123 (1324), 103-107.

Harden, K.P., Hill, J.E., Turkheimer, E., Emery, R.E. 2008. Gene-environment correlation and interaction in peer effects on adolescent alcohol and tobacco use. Behavior Genetics 38(4), 339-347.

Hellemans, K.G., Sliwowska, J.H., Verma, P., Weinberg, J. 2010. Prenatal alcohol exposure: Fetal programming and later life vulnerability to stress, depression and anxiety disorders. Neuroscience and Biobehavioral Reviews 34(6), 791-807.

Hillbom, M., Saloheimo, P., Juvela, S. 2011. Alcohol consumption, blood pressure and the risk of stroke. Current Hypertension Reports 13(3), 208-213.

Humphrey, G., Casswell, S. Han, D.Y. 2003. Alcohol and injury among attendees at a New Zealand emergency department. New Zealand Medical Journal 116(1168).

Irving, H.M., Samokhvalov, A.V., Rehm, J. 2009. Alcohol as a risk factor for pancreatitis: a systematic review and meta-analysis. JOP: Journal of the Pancreas (Online) 10(4), 387-392.

Kendler, K.S., Myers, J., Dick, D., Prescott, C.A. 2010. The relationship between genetic influences on alcohol dependence and on patterns of alcohol consumption. Alcoholism: Clinical and Experimental Research 34(6), 1058-1065.

Kypri, K. 2003. Maori/non-Maori alcohol consumption profiles: implications for reducing health inequalities. New Zealand Medical Journal 116(1184).

Kypri, K., Maclennan, B., Langley, J.D. and Connor, J.L. 2011. Commentary: *The Alcohol Reform Bill*: More tinkering than reform in response to the New Zealand public's demand for better liquor laws. Drug and Alcohol Review 30(4), 428-433.

Laixuthai, A., Chaloupka, F. 1993. Youth alcohol use and public policy. Contemporary Economic Policy 11(4), 70-81.

Laslett, A.M., Room, R., Ferris, J., Wilkinson, C., Livingston, M., Mugavin, J. 2011. Surveying the range and magnitude of alcohol's harm to others in Australia. Addiction 106(9), 1603-1611.

Latendresse, S.J., Rose, R.J., Viken, R.J., Pulkkinen, L., Kaprio, J., Dick, D.M. 2010. Examining the etiology of associations between perceived parenting and adolescents' alcohol use: common genetic and/or environmental liabilities? Journal of Studies on Alcohol and Drugs 71(3), 313-325.

Law Commission. 2009. Alcohol in Our Lives: An issues paper on the reform of New Zealand's liquor laws. Wellington: Law Commission.

Law Commission. 2010. Alcohol in Our Lives: Curbing the Harm. Wellington: Law Commission.

Lejuez, C.W., Magidson, J.F., Mitchell, S.H., Sinha, R., Stevens, M.C., de Wit, H. 2010. Behavioral and biological indicators of impulsivity in the development of alcohol use, problems, and disorders. Alcoholism: Clinical and Experimental Research 34(8), 1334-1345.

Lonnroth, K., Williams, B.G., Stadlin, S., Jaramillo, E., Dye, C. 2008. Alcohol use as a risk factor for tuberculosis - a systematic review. BMC Public Health 8(289).

Maclennan, B. 2010. Local government alcohol policy : community sentiment and the policy development process. PhD thesis. Dunedin: University of Otago.

Maldonado-Devincci, A.M., Badanich, K.A., Kirstein, C.L. 2010. Alcohol during adolescence selectively alters immediate and long term behavior and neurochemistry. Alcohol 44(1), 57-66.

McCarver, D.G., Thomasson, H.R., Martier, S.S., Sokol, R.J., Li, T-K. 1997. Alcohol dehydrogenase-2\*3 allele protects against alcohol-related birth defects among African Americans. Journal of Pharmacology & Experimental Therapeutics 283(3), 1095-1101.

McCreanor, T., Moewaka Barnes, H., Kaiwai, H., Borrell, S., Gregory, A. 2008. Creating intoxigenic environments: marketing alcohol to young people in Aotearoa New Zealand. Social Science and Medicine 67(6), 938-946.

Ministry of Health. 2009. Alcohol Use in New Zealand: Key Results of the 2007/08 New Zealand Alcohol and Drug Use Survey. Wellington: Ministry of Health. Available: http://www.moh.govt.nz/moh.nsf/pagesmh/9626/\$File/alcohol-use-in-nz-0708-survey.pdf Accessed 29.11.11.

Ministry of Justice. 2007. The sale and supply of alcohol to under 18 year olds in New Zealand: a systematic overview of international and New Zealand literature. Final Report. Wellington: Ministry of Justice. Available: <u>http://www.justice.govt.nz/publications/global-publications/t/the-sale-and-supply-of-alcohol-to-under-18-year-olds-in-new-zealand-a-systematic-overview-of-international-and-new-zealand-literature-final-report/11-international-policy-and-legislative-interventions Accessed 14.12.11</u>

Ministry of Social Development. 2010. *Learning from Tragedy: Homicide within Families in New Zealand 2002-2006*. Wellington: Ministry of Social Development. Available: <u>http://www.msd.govt.nz/about-msd-and-our-work/publications-resources/research/learning-from-tragedy/index.html</u> Accessed 30.11.11.

Ministry of Transport. 2010. *Alcohol/drugs Crash Factsheet*. Available at: <u>http://www.transport.govt.nz/research/Documents/Alcohol-drugs-crash-fact-sheet-2010.pdf</u>

National Business Review. 2011. Strong liquor exports rise 28%. Available: <u>http://www.nbr.co.nz/article/strong-liquor-exports-rise-28-ng-98178</u> Accessed 2.3.12.

National Drug Policy New Zealand. 2010. Alcohol – legislation. Available: <u>http://ndp.govt.nz/moh.nsf/indexcm/ndp-alcohol-legislation</u> Accessed 16.12.11.

New Zealand Drug Foundation. undated. Alcohol: law and penalties. Available: <u>http://www.drugfoundation.org.nz/alcohol/law-and-penalities</u> Accessed 14.12.11

New Zealand Transport Agency. 2011. Alcohol and drugs limits. Available: <u>http://www.nzta.govt.nz/resources/roadcode/about-limits/alcohol-and-drugs-limits.html</u> Accessed 16.12.11.

O'Connell, H., Lawlor, B.A. 2005. Recent alcohol intake and suicidality: a neuropsychological perspective. Irish Journal of Medical Science 174(4), 51-54

Oei, T., Morawska, A. 2004. A cognitive model of binge drinking: the influence of alcohol expectancies and drinking refusal self-efficacy. Addictive Behaviors 29(1), 159-179.

Ogden, G.R. 2005. Alcohol and oral cancer. Alcohol 35(3), 169-173.

O'Malley P.M. and Wagenaar A.C. 1991. Effects of minimum drinking age laws on alcohol use, related behaviors and traffic crash involvement among American youth: 1976-1987. Journal of Studies on Alcohol 52(5), 478-91.

Oze, I., Matsua, K., Wakai, K., Nagata, C., Mizoue, T., Tanaka, K. et al. 2011. Alcohol drinking and esophageal cancer risk: an evaluation based on a systematic review of epidemiologic evidence among the Japanese population. Japanese Journal of Clinical Oncology 41(5), 677-692.

Pitkanen, T., Kokko, K., Lyyra, A., Pulkkinen, L. 2008. A developmental approach to alcohol drinking behaviour in adulthood: a follow-up study from age 8 to 42. Addiction 103(Suppl 1), 48-68.

Panza, F., Capurso, C.,D'Introno, A., Colacicco, A.M., Frisardi, V., Lorusso, M., et al. 2009. Alcohol drinking: cognitive functions in older age, predementia, and dementia syndromes. Journal of Alzheimer's Disease 17(1), 7-31.

Rehm, J., Taylor, B., Mahapatra, S., Irving, H., Baliunas, J., Roerecke, M. 2010. Alcohol as a risk factor for liver cirrhosis: a systematic review and meta-analysis. Drug and Alcohol Review 29(4), 437-445.

Rimm, E.B., Williams, P., Fosher, K., Criqui, M., Stampfer, M.J. 1999. Moderate alcohol intake and lower risk of coronary heart disease: meta-analysis of effects on lipids and haemostatic factors. BMJ 319(7224), 1523-1528.

Roerecke, M., Rehm, J. 2010. Irregular heavy drinking occasions and risk of ischemic heart disease: a systematic review and meta-analysis. American Journal of Epidemiology 171(6), 633-644.

Samokhvalov, A.V., Irving, H.M., Rehm, J. 2010. Alcohol consumption as a risk factor for pneumonia: a systematic review and meta-analysis. Epidemiology and Infection 138(12), 1789-1795.

Samokhvalov, A.V., Irving, H., Mohapatra, S., Rehm, J. 2010. Alcohol consumption, unprovoked seizures and epilepsy: a systematic review and meta-analysis. Epilepsia 51(7), 1177-1184.

Sampson, H.W. 2002. Alcohol and other factors affecting osteoporosis risk in women. Alcohol Research and Health 26 (4), 292-298.

Sellman, J., Robinson, G., Beasley, R. 2009. Should ethanol be scheduled as a drug of high risk to public health? Journal of Psychopharmacology 23(1), 94-100.

Singh, J.A., Reddy, S.G., Kundukulam, J. 2011. Risk factors for gout and prevention: a systematic review of the literature. Current Opinion in Rheumatology 23(2), 192-202.

Slack, A., Nana, G., Webster, M., Stokes, F., Wu, J. 2009. *Costs of Harmful Alcohol and other Drug Use*. Wellington: Business and Economic Research Limited.

Sloane, K., Gunasekara, F, Wilson, N. 2011. Consistently cheap alcohol: national data on discounts for an 8-week period [Letter]. New Zealand Medical Journal 124 (1329), 104-108.

Spicer, R.S., Miller, T.R., Smith, G.S. 2003. Worker substance use, workplace problems, and the risk of occupational injury: a matched case-control study. Journal of Studies on Alcohol 64(4), 570-578.

Statistics New Zealand. 2011. Hot off the Press: Alcohol available for consumption: Year ended December 2010. Wellington: Statistics New Zealand. Available: http://www.stats.govt.nz/browse\_for\_stats/industry\_sectors/alcohol\_and\_tobacco\_availability/Alco\_hol-available-for-consumption\_HOTPYeDec10.aspx\_Accessed 16.12.11.

Stephenson, R. 2009. *National Alcohol Assessment*. Wellington: New Zealand Police. Available: <u>http://www.police.govt.nz/sites/default/files/Police-National-Alcohol-Assessment.pdf</u> Accessed 30.11.11.

Streissguth, A. P., Bookstein, F. L., Barr, H. M., Sampson, P.D., O'Malley, K., and J.K. Young. 2004. Risk factors for adverse life outcomes in foetal alcohol syndrome and foetal alcohol effects. Developmental and Behavioural Paediatrics 25 (4): 228-238.

Swann, A. 2009. Wine export growth outshines all others. National Business Review April 30. Available: <u>http://www.nbr.co.nz/article/wine-export-growth-outshines-all-others-101767</u> Accessed 2.3.12

Wagenaar, A.C., Wolfson, M. 1994. Enforcement of the legal minimum drinking age in the United States. Journal of Public Health Policy 15(1), 37-53.

Weir, R.P., Day, P., Ali, W. 2007. Risk factors for breast cancer in women: a systematic review of the literature. NZHTA Report 10(2), 1-361.

Wellington City Council. Undated. Sale of Liquor Act definitions. Available: <u>http://www.wellington.govt.nz/services/liquor/policies/actdefinitions.html</u> Accessed 29.11.11.

Winstanley, E.L., Steinwachs, D.M., Ensminger, M.E., Latkin, C.A., Stitzer, M.L. Olsen, Y. 2008. The association of self-reported neighborhood disorganization and social capital with adolescent alcohol and drug use, dependence and access to treatment. Drug and Alcohol Dependence 92(1-3), 173-182.

Witt, E.D. 2010. Research on alcohol and adolescent brain development: opportunities and future directions. Alcohol 44(1), 119-124.

World Health Organisation. 2011. Global status report on alcohol. Geneva, WHO. Available: <u>http://www.who.int/substance\_abuse/publications/global\_alcohol\_report/msbgsruprofiles.pdf</u> Accessed 8.11.11

World Health Organisation. 2007. WHO Expert Committee on problems related to alcohol consumption. Geneva, Switzerland: World Health Organisation.

Yeomans, M.R. 2010. Alcohol, appetite and energy balance: is alcohol intake a risk factor for obesity? Physiology and Behavior 100(1), 82-89.

Young-Wolff, K.C., Enoch, M-A., Prescott, C.A. 2011. The influence of gene-environment interactions on alcohol consumption and alcohol use disorder: a comprehensive review. Clinical Psychology Review 31 (5), 800-816.

Ziegler, D.W., Wang, C.C., Yoast, R.A., Dickinson, B.D., McCaffree, M.A., Robinowitz, C.B et al. 2005. The neurocognitive effects of alcohol on adolescents and college students. Preventive Medicine 40(1), 23-32.